## UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

Form 10-Q

[X] Quarterly report under Section 13 or 15(d) of the Securities Exchange Act of 1934

For The Quarterly Period Ended March 31, 1996

Commission File Nos. 0-9115 and 0-24494

MATTHEWS INTERNATIONAL CORPORATION (Exact Name of registrant as specified in its charter)

PENNSYLVANIA25-0644320(State or other jurisdiction of<br/>incorporation or organization)(I.R.S. Employer<br/>Identification No.)

TWO NORTHSHORE CENTER, PITTSBURGH, PA15212-5851(Address of principal executive offices)(Zip Code)

Registrant's telephone number, including area code (412) 442-8200

NOT APPLICABLE (Former name, former address and former fiscal year, if changed since last report)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15 (d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days.

Yes [X] No [ ]

The number of shares outstanding of each of the issuer's classes of common stock, as of the latest practicable date:

Class of Common Stock	Outstanding at April 30, 1996
Class A - \$1.00 par value	5,875,716 shares
Class B - \$1.00 par value	3,142,506 shares
PART I - FI	NANCIAL INFORMATION
MATTHEWS INTE	ERNATIONAL CORPORATION AND SUBSIDIARIES
CONSOLIDAT	TED BALANCE SHEET (UNAUDITED)
<table></table>	
<caption></caption>	
	March 31, 1996 September 30, 1995
<s></s>	<pre></pre>
ASSETS	
Current assets:	

Current assets.		
Cash and cash equivalents	\$ 14,237,869	\$ 39,204,010
Short-term investments	3,020,128	-
Accounts and notes receivable, net	27,280,451	28,515,610
Inventories:		

Materials and finished goods Labor and overhead in process Supplies Less LIFO reserve		812,178 8,907 (298,673)	l
Other current assets	11,956,736 966,916	10,341,823 1,174,7	96
Total current assets	57,462,100	79,236,	239
Investments Property, plant and equipment: Cost Less accumulated depreciation	33,797,171 63,072,483 (26,065,097)	- 62,429,5 (24,407,80	586 09)
Deferred income taxes and other assets Goodwill Total assets	37,007,386 s 10,99	38,021,777 97,415	15,588,221
Goodwill	10,504,170	5,360,139	)
Total assets	\$149,768,242	\$138,206,	376
LIABILITIES AND SHAKEHOLDER	439,6 4,179,501 6,389,876 3,241,119	597 5,181,9 5,181,9 8 7,94 1,162	433,465 954 14,824 5.805
Total current liabilities	22,980,100	22,910,	,020
Long-term debt Estimated cemetery and finishing costs Postretirement benefits Deferred revenue and other liabilities	48,668 s 3,00 20,139,350 1,539,	270,092 9,327 835 3	2 4,991,476 7,632 5,508,752
Shareholders' equity: Common stock: Class A, par value \$ Class B, par value \$1.00 Other shareholders' equity	1.00 5,863,296 3,200,916 92,986,750	4,009 4,840,597 77,948,054	9,753
	102,050,962	86,798,404	
Total liabilities and shareholders' equit	ty\$149,76	58,242	5138,206,376
/TABLE			

# MATTHEWS INTERNATIONAL CORPORATION AND SUBSIDIARIES CONSOLIDATED STATEMENT OF INCOME (UNAUDITED)

<TABLE>

<caption></caption>	Three Months March 31,		Six Months I March 31,	Ended
	1996 19 	995 19	996 1995	-
<s> Sales</s>	<c> <c> <c \$ 42,791,474</c </c></c>	e	> <c> \$ 83,976,824</c>	\$ 82,171,388
Cost of sales	23,819,715	23,180,292	2 46,421,717	44,902,530
Selling and administrative ex	penses 12,125	5,473 12,55	5,911 24,256	5,568 24,555,784
Operating profit	6,846,280	6 6,349,380	0 13,298,539	12,713,074
Other income (deductions), net	6,415,350	0 356,083	6,819,975	526,413
Interest expense	46,337	21,824	67,696	40,365

Income before incom	e taxes 13,21	5,299 6,68	83,639 20	),050,818 13,199,122
Income taxes	5,839,254			
Net income		\$ 4,080,628	\$ \$11,622,	034 \$ 7,990,293
Earnings per share	\$ .83	\$ .46 ==== =	\$ 1.31	\$ .90
Dividends per share	\$ .07	\$ .06	\$.14	\$ .12
Weighted average nu of common shares	mber			8,850,350 === ========
				N AND SUBSIDIARIES DWS (UNAUDITED)
<caption></caption>		Six Mont March		
		1996	1995	
<s> Cash flows from ope Net Income Adjustments to recor provided by operat Depreciation and an Deferred taxes Net increase in certa Decrease in other net Increase in estimate Decrease in deferred other liabilities Increase in postretir (Gain) loss on sale of (Gain) loss on sale of Gain on sale of subs Effect of exchange in Net cash provided Cash flows from inve Acquisition of subsit Purchases of investm Proceeds from sale of Proceeds from sale of Collections on loans</s>	ncile net incom ing activities: nortization ain working cap oncurrent assets d finishing and d revenue and e ement benefits of property, plan of investments sidiary rate changes on by operating ac esting activities erty, plant and diary, net of cas nents sals of property at or maturity of ir of subsidiary	<pre> <c> :: \$11,622 e to net cash 258, pital items cemetery cos expenses and (337,6 nt and equipm (9, operations ctivities : equipment sh acquired (42 v, 42 ivestments</c></pre>	 <c> 2,034 \$ 7,9 4,761,738 312 (344 (1,141,4; (199,198)) sts 211, 19) (39,2 411,718 nent (233 38,802 409,058) (329,8 5,653,513 (2,327, (3,482,3 2,099,157) 28,867 5,176, 13,070,853</c>	2,286,393 $4,600)$ $85) (3,400,848)$ $8,499$ $312 139,992$ $270)$ $733,894$ $5,145) 20,659$ $-$ $-$ $-$ $-$ $-$ $-$ $-$ $-$ $-$ $-$
Net cash used in ir		es		
Cash flows from fina Payments on long-te Proceeds from the sa Purchases of treasur Dividends paid	ncing activities rm debt ile of treasury s y stock	tock (1	),/31) (1,0	(207,909) - 61,647)
Net cash used in fi	nancing activiti	ies 	(2,432,673)	(1,269,556)
Effect of exchange ra	te changes on			

Effect of exchange rate changes on

cash and cash equivalents		94	4,09	4	(44,205)	
Net increase (decrease) in cash and c	ash	equivalent	s	\$(24,9	966,141)	\$ 4,660,703
Supplemental Cash Flow Informatio	n:					
Cash paid during the period for:						
Interest	\$	67,696	\$	40,3	65	
Income Taxes		5,269,2	205	5,6	540,294	
/TABLE						
MATTHEWS INTERNATI	ON/	AL CORPO	DRA	TION	I AND SU	<b>JBSIDIARIES</b>
NOTES TO CONSOLID	ATE	ED FINAN	ICIA	AL ST	ATEMEN	ITS
MARCH 31, 199	6		-			
	-					

## Note 1. Nature of Operations

Matthews International Corporation, founded in 1850 and incorporated in Pennsylvania in 1902, is a designer, manufacturer and marketer of custom-made products which are used to identify people, places, products and events. The Company's products and operations are comprised of three business segments: Bronze, Graphic Systems and Marking Products. The Bronze segment is a leading manufacturer of cast bronze memorial products used primarily in cemeteries. The Graphic Systems segment manufactures and provides custom identificationrelated products and services used by the corrugated packaging industry and the flexible packaging industry. The Marking Products segment designs, manufactures and distributes a wide range of equipment and consumables used by customers to mark or identify various consumer and industrial products, components and packaging containers. The Company has manufacturing facilities in the United States, Canada, Australia and Sweden as well as sales and distribution facilities in France and Germany.

## Note 2. Basis of Presentation

The accompanying consolidated financial statements have been prepared in accordance with generally accepted accounting principles for interim financial information for commercial and industrial companies and the instructions to Form 10-Q and Rule 10-01 of Regulation S-X. Accordingly, they do not include all of the information and footnotes required by generally accepted accounting principles for complete financial statements. In the opinion of management, all adjustments (consisting of normal recurring accruals) considered necessary for fair presentation have been included. Operating results for the three-month and six-month periods ended March 31, 1996 are not necessarily indicative of the results that may be expected for the fiscal year ending September 30, 1996. For further information, refer to the consolidated financial statements and footnotes thereto included in the Company's Annual Report on Form 10-K for the year ended September 30, 1995.

The preparation of financial statements in conformity with generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

## Note 3. Income Taxes

The income tax provision for the period is based on the effective tax rate expected to be applicable for the full year. The difference between the estimated effective tax rate of 42.0% and the Federal statutory rate of 35% primarily reflects the impact of state and foreign income taxes.

## MATTHEWS INTERNATIONAL CORPORATION AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS, continued MARCH 31, 1996

#### Note 4. Investments

On February 19, 1996, the Company acquired 40% of the common stock of Applied Technology Developments, Ltd., a British manufacturer of impulse ink-jet

printing equipment, at a cost of \$1,596,688. The investment has been recorded under the equity method of accounting.

All other investment securities are classified as available-for-sale and are recorded at market value at the consolidated balance sheet date. Short-term investments consist of securities with purchased maturities of over three months but less than one year. Accrued interest on all investment securities, including purchased interest, is also classified with short-term investments. Investments classified as non-current consist of securities with purchased maturities intended to range from one to five years.

Unrealized gains and losses on investment securities are included as a separate component of shareholders' equity. Realized gains and losses are based on the specific identification method and are recorded in other income. Bond premiums and discounts are amortized on the straight-line method which does not significantly differ from the interest method.

The amortized cost and market values of investment securities at March 31, 1996 were as follows:

$\begin{array}{c} (Amortized Unrealized Unrealized Market \\ Cost) Gains Losses Value \\ \hline \\ Corporate obligations 1,018,000 \\ \hline \\ Corporate obligations 18,668,773 \\ \hline \\ Corp$		Book Value	Gross	Gross	
Short-term investments: U.S. government and its agencies $$1,499,835$ $$780$ $$-$1,500,615$ Corporate obligations $1,018,000$ $  1,018,000$ Other $501,513$ $  501,513$ Total $$3,019,348$ $$780$ $$-$3,020,128$ Investments: U.S. government and its agencies $$14,003,340$ $$ $295,504$ $$13,707,836$ Corporate obligations $18,668,773$ $ 241,953$ $18,426,820$		(Amortized U	nrealized	Unrealized	Market
U.S. government and its agencies $$1,499,835$ $$780$ $$-$1,500,615$ Corporate obligations $1,018,000$ 1,018,000 Other 501,513 501,513 Total $$3,019,348$ $$780$ $$-$3,020,128$ ====== ==============================		Cost) Gai	ins Lo	osses Va	lue
U.S. government and its agencies $$1,499,835$ $$780$ $$-$1,500,615$ Corporate obligations $1,018,000$ 1,018,000 Other 501,513 501,513 Total $$3,019,348$ $$780$ $$-$3,020,128$ ====== ==============================					
its agencies \$ 1,499,835 \$ 780 \$ - \$ 1,500,615 Corporate obligations 1,018,000 1,018,000 Other 501,513 501,513 Total \$ 3,019,348 \$ 780 \$ - \$ 3,020,128 ======= =============================	Short-term invest	ments:			
Corporate obligations 1,018,000 1,018,000 Other 501,513 501,513 Total \$3,019,348 \$780 \$ - \$3,020,128 	U.S. governmen	t and			
Corporate obligations 1,018,000 1,018,000 Other 501,513 501,513 Total \$3,019,348 \$780 \$ - \$3,020,128 	its agencies	\$ 1,499,835	\$ 780	\$ - \$	1,500,615
Total       \$ 3,019,348 \$ 780 \$ - \$ 3,020,128	Corporate obliga	ations 1,018,	.000 -		1,018,000
Investments: U.S. government and its agencies \$14,003,340 \$ - \$295,504 \$13,707,836 Corporate obligations 18,668,773 - 241,953 18,426,820	Other	501,513	-	- 501,	513
Investments: U.S. government and its agencies \$14,003,340 \$ - \$295,504 \$13,707,836 Corporate obligations 18,668,773 - 241,953 18,426,820					
U.S. government and its agencies \$14,003,340 \$ - \$295,504 \$13,707,836 Corporate obligations 18,668,773 - 241,953 18,426,820	Total	\$ 3,019,348	\$ 780	\$ - \$3	,020,128
U.S. government and its agencies \$14,003,340 \$ - \$295,504 \$13,707,836 Corporate obligations 18,668,773 - 241,953 18,426,820					
U.S. government and its agencies \$14,003,340 \$ - \$295,504 \$13,707,836 Corporate obligations 18,668,773 - 241,953 18,426,820					
its agencies \$14,003,340 \$ - \$295,504 \$13,707,836 Corporate obligations 18,668,773 - 241,953 18,426,820	Investments:				
Corporate obligations 18,668,773 - 241,953 18,426,820	U.S. governmen	t and			
	its agencies	\$14,003,340	) \$ -	\$295,504	\$13,707,836
	Corporate obliga	ations 18,668	,773	- 241,9	53 18,426,820
Total \$32,737,940 \$ - \$537,457 \$32,200,483	Total	\$32,737,940	\$ -	\$537,457	\$32,200,483

## MATTHEWS INTERNATIONAL CORPORATION AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS, continued MARCH 31, 1996

## Note 5. Acquisition

On March 25, 1996, Matthews International Corporation acquired the stock of Industrial Equipment and Engineering Company ("IEEC"), for 213,862 shares of Matthews Class A common stock (valued at \$5,400,000) and \$3,600,000 cash. The acquisition was consummated through a statutory merger between a wholly-owned subsidiary of Matthews and IEEC (including a real estate holding corporation related to IEEC), with the Matthews subsidiary as the surviving corporation. In addition, Matthews' wholly-owned subsidiary executed employment agreements with the two shareholders of IEEC pursuant to which performance-based incentive compensation would be payable to such shareholders if the cumulative pre-tax earnings of the merged business for the five-year period beginning April 1, 1996 exceeds \$8 million, which amount is significantly greater than recent years profit levels. Matthews has accounted for this acquisition using the purchase method and, accordingly, has recorded the acquired assets and liabilities at their estimated fair values at the date of acquisition. The excess of the purchase price over the fair value of the net assets was recorded as goodwill to be amortized on a straight-line basis over 20 years.

## Note 6. Disposition

On January 5, 1996, Matthews International Corporation sold for \$13,100,000 cash its cemetery and mortuary facility (Sunland Memorial Park, Inc.) in Sun City, Arizona to Service Corporation International. Matthews recorded a pretax gain of \$9.4 million on the sale which was recorded in other income. Sunland Memorial Park, Inc., which was purchased in 1982, was the only such facility owned by the Company. The facility had sales in fiscal year 1995 of approximately \$5 million, representing about 3 percent the consolidated sales of the Company.

Note 7. Supplemental Cash Flow Information

On March 25, 1996, the Company issued 213,862 shares of authorized Class A common stock, with a value of \$5.4 million, in connection with the acquisition of IEEC (See Note 5).

## MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

## **Results of Operations**

The following table sets forth certain income statement data of the Company expressed as a percentage of net sales for the periods indicated.

	Six months end March 31,	led Years en September 3	
	1996 1995	1995 1994	1993
Sales	100.0% 100	0.0% 100.0% 1	00.0% 100.0%
Gross profit	44.7 45	.4 44.8 45.1	42.4
Operating profit	15.8 1	5.5 14.7 15.	1 11.6
Income before incom	me taxes 23	3.9 16.1 15.0	14.9 11.0(1)
Net income	13.8 9	9.3 8.8	6.6(1)

(1) Excludes the cumulative effect of changes in accounting principles for the adoptions of SFAS No. 106 and SFAS No. 109.

Sales for the six months ended March 31, 1996 were \$84.0 million and were \$1.8 million, or 2.2%, higher than sales of \$82.2 million for the first six months of fiscal 1995. The increase for the first six months of fiscal 1996 reflected higher sales in all three of the Company's business segments. Bronze segment sales for the period were up 2% over fiscal 1995 despite the sale of Sunland Memorial Park, Inc. in January 1996 (see "Liquidity and Capital Resources") reflecting increases in both price and unit volume. Marking Products sales for the first six months of fiscal 1996 were up 3% over the same period a year ago reflecting higher demand in Europe and Australia. The segment's international sales increased 20% over the same period a year ago which more than offset a decline in North American sales volume. Graphic Systems sales for the first six months of fiscal 1996 were slightly higher than the first six months of fiscal 1995. However, sales for the second quarter of fiscal 1996 improved significantly from the 1996 first quarter. Sales for this segment have been adversely impacted, beginning in the third quarter of fiscal 1995, from the postponement by many customers of printing plates purchases in an attempt to offset increased costs for linerboard.

Gross profit for the six months ended March 31, 1996 was \$37.6 million, or 44.7% of sales, compared to \$37.3 million, or 45.4%, for the first six months of fiscal 1995. The increase in gross profit of \$286,000, or 0.8%, was attributable principally to the Marking Products segment. International gross profit improved with higher sales volume and the current period also benefitted from a reduction in lower margin sales in Italy. Although sales for the Bronze segment increased during the current period, higher material costs (principally bronze ingot) resulted in a relatively consistent gross profit amount for the segment compared to the same period a year ago. Graphic Systems gross profit and gross profit as a percent of sales for the first six months of fiscal 1996

Selling and administrative expenses for the six months ended March 31, 1996 were \$24.3 million, representing a decrease of \$300,000, or 1.1%, from \$24.6 million for the first six months of fiscal 1995. The reduction in selling and administrative costs for the first six months of fiscal 1996 reflects the absence of Sunland Memorial Park, Inc., which was sold in January 1996. In addition, selling costs declined in the Marking Products segment mainly due to the discontinuance of the Company's Italian operations effective November 1, 1995.

Operating profit for the six months ended March 31, 1996 was \$13.3 million and was \$585,000, or 4.6%, higher than operating profit of \$12.7 million for the first six months of fiscal 1995. The operating profit for the first six months of fiscal 1996 principally reflected improvements in the Marking Products segment, related to increased sales and related gross profit of its international operations. Operating profit for the Bronze segment was slightly ahead of last year as a result of an increase in sales combined with reductions in selling and administrative costs from the sale of Sunland. Graphic Systems operating profit was slightly below the prior year level primarily reflecting a constant level of sales combined with higher operating expenses.

Interest expense for the six months ended March 31, 1996 was \$68,000, compared to \$40,000 for the first six months of fiscal 1995. Interest expense principally relates to the Company's capital lease obligations.

Other income (deductions), net for the six months ended March 31, 1996 was \$6.8 million compared to \$526,000 for the first six months of fiscal 1995. Other income for the first six months of fiscal 1996 reflected a \$9.4 million pre-tax gain on the sale of Sunland Memorial Park, Inc. This gain was partially offset by the write-off of the remaining goodwill (\$2.3 million) with respect to the Company's investment in its Swedish subsidiary which manufactures drop-on-demand ink-jet printing equipment and a charge for certain other non-operating expenses during the current period. Investment income for the first six months of fiscal 1996 was \$1.1 million compared to \$606,000 for the same period a year ago.

The Company's effective tax rate for the six months ended March 31, 1996 was 42.0%, compared to 38.4% for the year ended September 30, 1995. The higher effective tax rate for fiscal 1996 is primarily the result of a increase in the estimated impact of foreign income taxes, mainly in Sweden and Germany, on the Company's consolidated tax position. The difference between the Company's effective tax rate and the Federal statutory rate of 35% primarily reflects the impact of state and foreign income taxes.

#### Liquidity and Capital Resources

Net cash provided by operating activities was \$5.7 million for the six months ended March 31, 1996, compared to \$7.6 million for the first six months of fiscal 1995. Operating cash flow for the first six months of fiscal 1996 reflected the Company's net income for the period of \$11.6 million adjusted to exclude the gain on the sale of Sunland and non-cash depreciation and amortization (including the write-off of goodwill related to the Company's Swedish subsidiary). Cash flow from operations for the first six months of fiscal 1995 principally reflected the Company's net income for the period. Cash used in investing activities was \$28.3 million for the six months ended March 31, 1996 compared to \$1.6 million for the same period a year ago. Investing activities for the first six months of fiscal 1996 included net investments of \$35.3 million in short-term and intermediate-term securities of the U.S. government and its agencies and corporate obligations. These investments are designed to improve the investment rate of return on the Company's excess cash position while maintaining a sufficient degree of liquidity for future cash needs. Investing activities also included the acquisition for \$1.6 million of 40% of the common stock of Applied Technology Developments, Ltd., a British manufacturer of impulse ink-jet printing equipment.

Capital expenditures for the six months ended March 31, 1996 amounted to

\$2.3 million, representing approximately the same level of capital expenditures for the first six months of fiscal 1995. Capital spending for property, plant and equipment has averaged approximately \$4.9 million for the last three fiscal years. The capital budget of the Company for fiscal 1996 is \$12.7 million. The Company expects to generate sufficient cash from operations to fund all anticipated capital spending projects.

On January 5, 1996, the Company sold for \$13.1 million cash its cemetery and mortuary facility (Sunland Memorial Park, Inc.) in Sun City, Arizona to Service Corporation International. The transaction resulted in a pre-tax gain of \$9.4 million. Sunland Memorial Park, Inc., which was purchased in 1982, was the only such facility owned by Matthews. The facility had sales in fiscal year 1995 of approximately \$5 million, representing about 3 percent the consolidated sales of the Company. The sale of Sunland Memorial Park, Inc. will permit Matthews International Corporation to concentrate on growing its core business profitability. The proceeds from this transaction will provide additional resources to fund internal and acquisition growth in the Company's related businesses.

On March 25, 1996, the Company acquired the stock of Industrial Equipment and Engineering Company, Inc. ("IEEC"), for 213,862 newly-issued shares of authorized Matthews Class A common stock (approximately \$5.4 million) and \$3.6 million cash. IEEC, headquartered in Orlando, Florida, is the leading North American manufacturer of cremation equipment and also a supplier of related cremation products. IEEC sales were approximately \$7.5 million for the year ended December 31, 1995 and consisted of about 70% in equipment, 15% in field repairs, and the remainder in cremation supply products. The merger with IEEC is expected to provide Matthews with the opportunity to further participate in the increasing cremation trend and expand its range of products and services to the death care industry.

Cash used in financing activities for the six months ended March 31, 1996 was \$2.4 million principally reflecting the Company's quarterly dividends of \$.07 per share, the purchase of treasury stock and capital lease payments. Cash used in financing activities in the first six months of fiscal 1995 was \$1.3 million consisting of dividends and capital lease payments. Dividends for each of the first two quarters of fiscal 1995 were \$.06 per share. The Company currently has available lines of credit of approximately \$11 million. There were no outstanding borrowings on any of the Company's lines of credit at March 31, 1996. As of such date, the Company's outstanding long-term debt, which consisted of capital lease obligations, was approximately \$500,000. At March 31, 1996 and September 30, 1995 and 1994, the Company's current ratio was 2.5, 3.5 and 2.9, respectively. The Company had cash and cash equivalents at March 31, 1996 and September 30, 1995 of \$14.2 million and \$39.2 million, respectively. Net working capital at March 31, 1996 was \$34.5 million. The reduction in the current ratio and cash and cash equivalents balance from September 30, 1995 reflects investments in securities with longer maturities. The Company believes that its current liquidity sources, combined with its operating cash flow and additional borrowing capacity, will be sufficient to meet its capital needs for the next 12 months.

## Other

On March 15, 1996, the Board of Directors elected David M. Kelly, the Company's President and Chief Executive Officer, as Board Chairman. William M. Hauber, the Company's previous Chairman who retired from active service as Chief Executive Officer on October 1, 1995, resigned from the Board of Directors following a distinguished 45-year career with Matthews concurrent with Mr. Kelly's election as Chairman.

Item 4. Submission of Matters to a Vote of Security Holders

The Annual Meeting of the Shareholders of Matthews International Corporation was held on February 16, 1996. Total shares eligible for vote at such meeting were:

Class A Common Stock (one vote per share)	5,020,250 shares
Class B Common Stock (ten votes per share)	3,830,100 shares

The matters voted upon at such meeting were as follows:

1. Election of Directors:

The following individuals were nominated for election to the Board of Directors for terms expiring at the Annual Meeting of Shareholders in the year as set forth below. The nominations were made by the Board of Directors and no other nominations were made by any shareholder. The nominees had currently been members of the Board of Directors at the date of the Annual Meeting.

		Votes	
	Term	Withh	old
Nominee	Expiration	For	Authority
D.M. Kelly	1999	35,708,244	106,365
J.L. Parker	1999	35,717,644	96,965

The terms of the following additional directors continued after the meeting: W.A. Coates, G.D. Barefoot, D.J. DeCarlo, W.M. Hauber, T.N. Kennedy, J.P. O'Leary, Jr. and W.J. Stallkamp. On March 15, 1996, Mr. Hauber resigned as a director.

## 2. Selection of Auditors:

The shareholders voted to ratify the appointment by the Board of Directors of Coopers & Lybrand as independent certified public accountants to audit the records of the Company for the year ending September 30, 1996.

Votes For:	35,353,474
Votes Against:	400,255
Abstaining:	60,880

Item 6. Exhibits and Reports on Form 8-K

## (a) Exhibits

The following Exhibits to this report are filed herewith or incorporated by reference:

Exhibit

No. Description

- 10.1 Capital Stock Purchase Agreement, Sunland Memorial Park, Inc., incorporated by reference to Exhibit No. 10.1 to Form 10-Q for the quarter ended December 31, 1995
- 10.2 Agreement and Plan of Merger, Industrial Equipment and Engineering Company, Inc., filed herewith
- 11 Computation of Earnings Per Share, filed herewith

Item 6. Exhibits and Reports on Form 8-K, continued

## (b) Reports on Form 8-K

A Form 8-K current report was filed by the Company on January 17, 1996 reporting the following under "Item 5 - Other Events:"

Matthews International Corporation (the "Company") announced on January 5, 1996 that it has sold for cash its cemetery and mortuary facility (Sunland Memorial Park, Inc.) in Sun City, Arizona to Service Corporation International (SCI). Sunland Memorial Park, Inc., which was purchased in 1982, was the only such facility owned by the Company. The facility had sales in fiscal year 1995 of approximately \$5 million, representing about 3 percent the consolidated sales of the Company.

Matthews International Corporation will realize a pre-tax gain of approximately \$10 million on this sale, which will be recorded in the

Company's second quarter (March 31, 1996) financial statements.

The sale of Sunland Memorial Park, Inc. will permit Matthews International Corporation to concentrate on growing its core business profitability. The proceeds from this transaction will provide additional resources to fund internal and acquisition growth in the Company's related businesses.

A Form 8-K current report was filed by the Company on March 11, 1996 reporting the following under "Item 5 - Other Events:"

On March 5, 1996, Matthews International Corporation ("Matthews") signed an agreement and plan of merger to acquire the stock of Industrial Equipment and Engineering Company ("IEEC"), for 213,862 shares of Matthews Class A common stock and \$3,600,000 cash. The agreement provides that a wholly-owned subsidiary of Matthews will, on the closing date, merge with IEEC and its related real estate holding corporation. The Matthews subsidiary will be the surviving corporation. In addition, on the closing date, Matthews' wholly-owned subsidiary will execute employment agreements with the two shareholders of IEEC pursuant to which performance-based incentive compensation would be payable to such shareholders if the cumulative pre-tax earnings of the merged business for the five-year period beginning April 1, 1996 exceeds \$8 million, which amount is significantly greater than recent years profit levels.

A closing of the transaction, subject to normal due diligence review, is expected to occur prior to March 31, 1996.

IEEC, headquartered in Orlando, Florida, is the leading North American manufacturer of cremation equipment and also a supplier of related cremation products. The business was started in 1946 and was purchased by the current shareowners, P. Rahill and K. Robinson, in 1984. IEEC sales were approximately \$7.5 million for the year ended December 31, 1995 and consisted of about 70% in equipment, 15% in field repairs, and the remainder in cremation supply products. IEEC serves over 2,000 accounts worldwide.

The merger with IEEC is expected to provide Matthews with the opportunity to further participate in the increasing cremation trend and expand its range of products and services to the death care industry.

## SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

# MATTHEWS INTERNATIONAL CORPORATION (Registrant)

Date 5/10/96

E.J. Boyle

E.J. Boyle, Vice President, Accounting & Finance Date 5/10/96

J.L. Parker

J. L. Parker, Senior Vice President, General Counsel and Secretary

## AGREEMENT AND PLAN OF MERGER

By and Among

INDUSTRIAL EQUIPMENT AND ENGINEERING COMPANY, INC., a Florida corporation

FLORIDA DEVELOPMENT CORPORATION, a Florida corporation

MATTHEWS INTERNATIONAL CORPORATION, a Pennsylvania corporation

INDUSTRIAL EQUIPMENT AND ENGINEERING COMPANY, INC., a Delaware corporation

PAUL RAHILL, an individual

AND

KENNETH ROBINSON, an individual

Dated as of March 5, 1996

TABLE OF CONTENTS

ARTICLE I The Merger; The Closing

1.01. Terms of Merger

- 1.02. Surviving Corporation
- 1.03. Treatment of Shares
- 1.04. Effective Time
- 1.05. Certificate of Incorporation
- 1.06. By-Laws
- 1.07. Directors and Officers
- 1.08. Effect of Merger
- 1.09. Plan of Merger
- 1.10. The Closing

- 2.01. Conditions to the Obligations of the Acquiror
- (a) Certain Pre-Closing Notices and Other Items
- (b) Accuracy of Representations and Warranties; Performance of Covenants
- (c) Litigation
- (d) Material Adverse Change
- (e) Due Diligence
- (f) Title Insurance
- (g) Phase I Environmental Survey
- (h) Financial Statements
- (I) Employment and Non-Competition Agreement
- (j) Legal Opinion
- (k) Secretary s Certificate
- (1) Third-Party Consents and Similar Matters
- 2.02. Conditions to the Obligations of FDC, IEEC and the Shareholders
- (a) Accuracy of Representations and Warranties;
- Performance of Covenants (b) Legal Opinion
- (c) Employment Agreements
- (d) Third-Party Consents and Similar Matters

## ARTICLE III Pre-Closing Covenants

# 3.01. Covenant of Cooperation

- 3.02. Exclusivity
- 3.03. Operation and Preservation of Business
- 3.04. Confidentiality
- 3.05. Disclosure

# ARTICLE IV

## Post-Closing Covenants

- 4.01. Further Assurances
- 4.02. Payment of Taxes
- 4.03. Matthews Stock
- 4.04. Legends
- 4.05. Employee Benefits

## ARTICLE V Representations and Warranties

- 5.01. Representations and Warranties of FDC, IEEC and the Shareholders
- (a) Organization, Qualification and Corporate Power
- (b) Authorization of Transaction; Enforceability
- (c) No Contravention
- (d) Capitalization; Ownership of Shares
- (e) Subsidiaries
- (f) Financial Statements; Liabilities; Guarantees; No Material Adverse Change
- (g) Real Property; Leases; Assets; Liens; Condition of Properties
- (h) Accounts Receivable, Reimbursement Claims
- (I) Contracts
- (j) Intellectual Property
- (k) Litigation; Liabilities; Legal Compliance
- (l) Insurance
- (m) Tax Matters
- (n) Employees
- (o) Employee Benefits
- (p) Employee Severance Claims
- (q) Environment, Health and Safety
- (r) Governmental Licenses and Permits
- (s) Powers of Attorney; Bank Accounts
- (t) Certain Business Relationships with FDC or IEEC
- (u) Customers and Suppliers

- (v) Inventory
- (w) Warranty Expense
- (x) Brokers' Fees
- (y) Full Disclosure
- (z) Acquisition Entirely for Own Account
- (aa) Disclosure of Information
- (bb) Restricted Securities
- (cc) Accredited Investor
- (dd) Blue Sky
- 5.02. Representations and Warranties of the Acquiror and Matthews
- (a) Organization and Corporate Power
- (b) Authorization of Transaction
- (c) No Contravention
- (d) Brokers' Fees
- (e) Matthews' Exchange Act Filings

## ARTICLE VI

## Remedies for Breaches of this Agreement

- 6.01. Survival of Representations, Warranties and Covenants
- 6.02. Indemnification Provisions for Benefit of the Acquiror
- 6.03. Indemnification Provisions for Benefit of the Shareholders
- 6.04. Notice of Claim for Indemnification
- 6.05. Matters Involving Third Parties
- 6.06. Treatment of Indemnification Payments
- 6.07. Other Indemnification Provisions
- 6.08. Set-Off

## ARTICLE VII Miscellaneous

- 7.01. Further Assurances
- 7.02. Expenses
- 7.03. Press Releases and Public Announcements
- 7.04. Notices
- 7.05. No Third-Party Beneficiaries
- 7.06. Dispute Resolution
- (a) Specific Performance
- (b) Governing Law
- (c) Submission to Jurisdiction
- 7.07. Amendments and Waivers
- 7.08. Succession and Assignment
- 7.09. Termination
- 7.10. Matters of Construction, Interpretation and the Like
- (a) Construction
- (b) Severability
- (c) Entire Agreement; Counterparts

## ARTICLE VIII Definitions

Adverse Consequences Affiliate Affiliated Group Agreement Basis Closing Closing Date Code Confidential Information Disclosure Schedule Effective Time Employee Benefit Plan Employee Pension Benefit Plan Employee Welfare Benefit Plan Employment Agreements Environmental, Health and Safety Law ERISA Extremely Hazardous Substance Fiduciary Financial Statements Governmental Body Indemnified Party Indemnifying Party Intellectual Property Internal Revenue Service Knowledge Law Lien Liability Matthews Matthews Stock Merger Most Recent Balance Sheet

Multi employer Plan Ordinary Course of Business PBGC Permitted Liens Person Plan of Merger Prohibited Transaction Purchase Price Real Property Reportable Event Securities Act Shareholders Shares Subsidiary Surviving Corporation Tax; Taxes Tax Return Third-Party Claim

Exhibit A Form of Employment Agreement Exhibit B Registration Rights Agreement Exhibit C Financial Statements dated 12-31-95

Disclosure Schedule

Section 5.01(a)	Organization; Qualification; Directors and Officers
Section 5.01 (c)	No Contravention
Section 5.01(g)(I)	Real Property
Section 5.01(g)(ii)	
Section 5.01(h)	Accounts Receivable Aging
Section 5.01 (i)	Contracts
Section 5.01(j)(iii)	Intellectual Property
Section 5.01 (k)(i)	Litigation
Section 5.01(1)	Insurance
Section 5.01(m)	Tax matters
Section 5.01 (n)	Employees
Section 5.01(0)(i)	Employee Benefits
Section 5.01(r)	Licenses and Permits
Section 5.01 (s)	Bank Accounts
Section 5.01(t)	Certain Business Relationships
Section 5.01(u)	Customers and Suppliers
Section 5.01(w)	Warranty Expense

## AGREEMENT AND PLAN OF MERGER

THIS AGREEMENT AND PLAN OF MERGER (this "Agreement") dated as of March 5, 1996 by and among Florida Development Corporation, a Florida corporation ("FDC"), Industrial Equipment and Engineering Company, Inc., a Florida corporation ("IEEC"), Matthews International Corporation, a Pennsylvania corporation, and the sole shareholder of the Acquiror ("Matthews"), Industrial Equipment and Engineering Company, Inc., a Delaware corporation (the "Acquiror"), Paul Rahill, an individual ("Rahill") and Kenneth Robinson, an individual ("Robinson," and along with Rahill, the "Shareholders").

## WITNESSETH:

WHEREAS, all the issued and outstanding shares of capital stock of FDC and IEEC are owned by the Shareholders;

WHEREAS, the respective Boards of Directors of FDC, IEEC and the Acquiror deem it advisable for the general welfare and advantage of each and of their respective shareholders that FDC and IEEC merge into the Acquiror pursuant to this Agreement and the applicable laws of the States of Delaware and Florida;

WHEREAS, the Board of Directors of each of FDC, IEEC, and the Acquiror has approved this Agreement and the Plan of Merger contemplated hereby;

WHEREAS, the shareholders of each of FDC, IEEC and the Acquiror have approved the Plan of Merger contemplated hereby;

WHEREAS, this Merger is intended to be a tax-free reorganization under the terms of Section 368(a)(2)(D) of the Code;

NOW, THEREFORE, in consideration of the premises and of the mutual covenants and conditions herein contained, and intending to be legally bound hereby, the parties have agreed as follows:

## ARTICLE I The Merger; The Closing

1.01. Terms of Merger. At the Effective Time, FDC and IEEC shall be merged with and into the Acquiror pursuant to the provisions of the Florida 1989 Business Corporation Act and the provisions of the General Corporation Law of Delaware (the "Merger").

1.02. Surviving Corporation. The corporation surviving the Merger shall be the Acquiror (in its capacity as such surviving corporation, being sometimes referred to herein as the "Surviving Corporation").

1.03. Treatment of Shares. At the Effective Time, (a) the shares of capital stock of the Acquiror issued and outstanding immediately prior to the Merger shall remain outstanding, without change therein by reason of the Merger; (b) each share of capital stock of FDC issued and outstanding immediately prior to the Merger, by virtue of the Merger, and without any action on the part of the Shareholders, shall be converted into the right to receive (upon surrender of the certificate(s) representing the same) the following from the Surviving Corporation: (i) cash in the amount of \$2,000, without interest, and (ii) 118.81 shares of Class A Common Stock, par value \$1.00 per share, of Matthews (the "Matthews Stock"); (c) each share of capital stock of IEEC issued and outstanding immediately prior to the Merger, by virtue of the Merger, and without any action on the part of the Shareholders, shall be converted into the right to receive (upon surrender of the certificate(s) representing the same) the following from the Surviving Corporation: (i) cash in the amount of \$3,200 without interest, and (ii) 190.099 shares of Matthews Stock; and (d) each share of capital stock held in the treasury of FDC and IEEC immediately prior to the Effective Time shall not be converted but shall automatically be canceled at the Effective Time.

1.04. Effective Time. If this Agreement is not terminated as contemplated by Section 7.09 hereof, the Merger shall become effective upon the date and time of the last to occur of the filing of (i) Articles of Merger in the Department of State of Florida and (ii) a Certificate of Merger in the office of the Secretary of State of Delaware (the "Effective Time").

1.05. Certificate of Incorporation. The Certificate of Incorporation of the Acquiror as in effect at the Effective Time, from and after the Effective Time and until further amended as provided by applicable law, shall be, and may be separately certified as, the Certificate of Incorporation of the Surviving Corporation.

1.06. By-Laws. The By-Laws of the Acquiror, as in effect at the Effective Time, shall be the By-Laws of the Surviving Corporation, to remain unchanged until amended in accordance with the provisions thereof and of applicable law.

1.07. Directors and Officers. At the Effective Time, the Board of Directors of the Surviving Corporation shall consist of those persons who were directors of the Acquiror immediately prior to the Effective Time, and the officers of the Surviving Corporation shall be the persons who were officers of the Acquiror immediately prior to the Effective Time, each such person to hold, in accordance with the By-Laws and at the pleasure of the Board of Directors of the Surviving Corporation, the same office or offices with the Surviving Corporation as he or she then held with the Acquiror.

1.08. Effect of Merger. At the Effective Time, all the property, rights, privileges, franchises, patents, trademarks, licenses, registrations and other assets of every kind and description of FDC and IEEC shall be transferred to, vested in and devolve upon the Acquiror without further act or deed and all property, rights, and every other interest of the Acquiror and FDC and IEEC shall be as effectively the property of the Acquiror as they were of the Acquiror and FDC and IEEC respectively. FDC and IEEC hereby agree from time to time, as and When requested by the Acquiror or by its successors or assigns, to execute and deliver or cause to be executed and delivered all such deeds and instruments and to take or cause to be taken such further or other action as the Acquiror may deem necessary or desirable in order to vest in and confirm to the Acquiror title to and possession of any property of FDC and IEEC acquired or to be acquired by reason of or as a result of the Merger herein provided for and otherwise to carry out the intent and purposes hereof and the proper officers and directors of FDC and IEEC and the proper officers and directors of the Acquiror are fully authorized in the name of FDC and IEEC or otherwise to take any and all such action.

1.09. Plan of Merger. The foregoing provisions of Article I may be excerpted and used as a "Plan of Merger" with respect to FDC, IEEC and the Acquiror for approval by their respective Boards of Directors and shareholders.

## 1.10. The Closing.

(a) The consummation of the Merger and the execution and delivery of the Employment Agreements shall constitute the Closing. The Closing shall take place at the offices of Matthews International Corporation, 2 NorthShore Center, Pittsburgh, Pennsylvania, at 10:00 a.m. Eastern time on March 26, 1996 or at such other time or place or on such other date as shall be mutually agreed by the parties hereto. The Closing shall be effective for tax and accounting purposes as of 12:01 a.m. on March 26, 1996.

(b) At the Closing (i) FDC, IEEC and the Shareholders shall take the actions, and deliver to the Acquiror the materials, referred to in Section 2.01 hereof, (ii) the Acquiror and Matthews shall take the actions and deliver to FDC, IEEC and the Shareholders the materials referred to in Section 2.02 hereof, (iii) the Acquiror and Matthews shall deliver to the Shareholders the consideration referred to in Section 1.03 hereof and (iv) the Articles of Merger shall be filed in Florida and the Certificate of Merger shall be filed in Delaware.

### ARTICLE II Conditions to Obligations to Close

2.01. Conditions to the Obligations of the Acquiror and Matthews. The obligations of the Acquiror and Matthews to take the actions to be taken by the Acquiror

and Matthews under Article I hereof are subject to the satisfaction, prior to or on the Closing Date (or, in the case of Section 2.01 (a), Section 2.01 (e) and Section 2.01 (h) hereof, the other date specified below), of the following conditions precedent:

(a) Certain Pre-Closing Notices and Other Items.

(i) No later than March 8, 1996 FDC, IEEC and the Shareholders shall have (A) identified those contracts to which FDC or IEEC is a party which require consent to assignment in connection with the Merger, (B) sent to the other party to each such contract a letter soliciting such party's consent to the assignment of such contract and delivered to the Acquiror copies of such letters and evidence of the delivery of such letters.

(ii) No later than March 8, 1996 FDC, IEEC and the Shareholders shall have (A) identified those licenses and permits issued to FDC or IEEC which require consent to transfer the same to in connection with the Merger, (B) sent to the issuer of each such license or permit a letter soliciting consent to the transfer thereof, (C) delivered to the Acquiror copies of such letters and evidence of the delivery of such letters, (D) sent to the appropriate parties notice of the proposed transfer of all other licenses and permits to be transferred at the Closing and (E) delivered to the Acquiror copies of all such notices and evidence of the delivery of such notices.

(iii) No later than March 8, 1996 FDC, IEEC and the Shareholders shall have delivered to the Acquiror a preliminary draft of the Disclosure Schedule.

(b) Accuracy of Representations and Warranties; Performance of Covenants. The representations and warranties of FDC, IEEC and the Shareholders in Article V shall have been true in all material respects when made and, in addition, shall be true in all material respects on and as of the Closing Date, with the same force and effect as though made on and as of the Closing Date. The covenants of FDC, IEEC and the Shareholders in Article III shall have been performed and complied with in all material respects. The Acquiror shall have received a certificate dated the Closing Date signed by FDC, IEEC and the Shareholders, in such detail as the Acquiror may reasonably request, certifying as to the satisfaction of the conditions in this Section 2.01(b).

(c) Litigation. No action, suit or proceeding shall be pending before any court, agency, department or commission wherein an unfavorable injunction, judgment, order, decree, ruling or charge would (i) prevent consummation of any of the transactions contemplated by this Agreement, (ii) cause any of such transactions to be rescinded after their consummation, (iii) adversely affect the right of FDC or IEEC to own its properties and to conduct its business or (iv) adversely affect the right of the

Acquiror to own its properties and to conduct its business. No such injunction, judgment, order, decree or ruling shall be in effect, whether or not final or appealable. No person shall have notified the Acquiror, FDC, IEEC or the Shareholders that such person is considering the institution of any such action, suit or proceeding.

(d) Material Adverse Change. Since December 31, 1995, neither FDC nor IEEC shall have suffered any change in its condition (financial or otherwise) or in its assets, liabilities (absolute or contingent) or business, except changes in the lawful and ordinary course of business, which, individually or in the aggregate, have not materially and adversely affected its business, property, assets, condition or prospects.

(e) Due Diligence. The Acquiror shall have concluded a due diligence review of FDC and IEEC and the results of such the due diligence review shall have been found to be satisfactory to the Acquiror in all respects, provided, however, that this condition shall be deemed to have been satisfied unless the Acquiror provides notice to IEEC within fourteen (14) days after the date the Schedules are delivered that such due diligence review was found to be unsatisfactory to the Acquiror in any respect in which event this Agreement shall terminate without liability to any party.

(f) Title Insurance. The Acquiror shall have received title insurance policies, or binding commitments to issue such policies, covering all of the Real Property in form and substance satisfactory to the Acquiror.

(g) [Deleted]

(h) [Deleted]

(i) Employment and Non-Competition Agreements. Each of the Shareholders shall have executed and delivered to the Acquiror an Employment and Non-Competition

Agreement in the form attached hereto as Exhibit A.

(j) Legal Opinion. The Acquiror shall have received from Lowndes, Drosdick, Doster, Kantor & Reed, P.A. counsel for FDC, IEEC and the Shareholders, an opinion dated the Closing Date, in form and substance reasonably satisfactory to Acquiror.

(k) Secretary's Certificate. Each of FDC and IEEC shall have executed and delivered to the Acquiror a Secretary's Certificate signed by the Secretary and the President of each of FDC and IEEC (i) certifying as to the office held and true signature of the officers of each of FDC and IEEC executing this Agreement, and (ii) certifying as to attached copies of the articles of incorporation and bylaws of each of FDC and IEEC and resolutions of the board of directors and the Shareholders authorizing the transactions contemplated by this Agreement.

(1) Third-Party Consents and Similar Matters. All authorizations, consents and approvals of Governmental Bodies or third parties, if any, necessary or advisable in order to satisfy the conditions in Section 2.01 (b) hereof shall have been obtained and copies of such executed authorizations, consents and approvals delivered to the Acquiror. Without limiting the generality of the foregoing, with respect to all contracts to which FDC or IEEC is a party as of the Closing Date, FDC or IEEC shall have obtained all necessary consents to assignment required under the terms of such contracts and shall have delivered to the Acquiror copies of such executed consents.

(m) Registration Rights Agreement. The Shareholders shall have executed and delivered to Matthews the Registration Rights Agreement in the form attached hereto as Exhibit B.

2.02. Conditions to the Obligations of FDC, IEEC and the Shareholders. The obligations of FDC, IEEC and the Shareholders to take the actions to be taken by FDC, IEEC and the Shareholders, respectively, at the Closing pursuant to this Agreement are subject to the satisfaction, prior to or on the Closing Date, of the following conditions precedent:

(a) Accuracy of Representations and Warranties; Performance of Covenants. The representations and warranties of the Acquiror and Matthews in Articles V shall have been true in all material respects when made and, in addition, shall be true in all material respects on and as of the Closing Date, with the same force and effect as though made on and as of the Closing Date. The covenants of the Acquiror and Matthews in Article III shall have been performed and complied with in all material respects. FDC, IEEC and the Shareholders shall have received a certificate dated the Closing Date signed by the Acquiror and Matthews certifying, in such detail as FDC, IEEC and the Shareholders may reasonably request, to the satisfaction of the conditions in this Section 2.02(a).

(b) Legal Opinion. FDC, IEEC and the Shareholders shall have received from James L. Parker, Esq., General Counsel of the Acquiror and Matthews, an opinion dated the Closing Date, in form and substance reasonably satisfactory to FDC, IEEC and the Shareholders.

(c) Employment Agreements. The Acquiror shall have executed and delivered to the Shareholders the Employment and Non-Competition Agreements in the form attached hereto as Exhibit A.

(d) Third-Party Consents and Similar Matters. All authorizations, consents and approvals of Governmental Bodies or third parties, if any, necessary or advisable in order to satisfy the conditions in Section 2.02(a) hereof shall have been obtained and copies of such executed authorizations, consents and approvals delivered to FDC or IEEC.

(e) Secretary's Certificate. Each of Acquiror and Matthews shall have executed and delivered to Shareholders a Secretary's Certificate signed by the Secretary and the President of each of Acquiror and Matthews (i) certifying as to the office held and true signature of the officers of each of Acquiror and Matthews executing this Agreement, and (ii) certifying as to the resolutions of the board of directors of Acquiror and Executive Committee of Matthews authorizing the transactions contemplated by this Agreement.

(f) Registration Rights Agreement. Matthews shall have executed and delivered to Shareholders the Registration Rights Agreement in the form attached hereto

## ARTICLE III Pre-Closing Covenants

3.01. Covenant of Cooperation. Until the Closing Date, the Acquiror, FDC, IEEC and the Shareholders shall use reasonable good faith efforts to take all actions necessary or advisable in order to consummate and make effective the transactions contemplated by this Agreement (including satisfaction, but not waiver, of the closing conditions set forth in Article II). FDC and IEEC shall give any notices to third parties and shall use its best efforts to obtain any third-party consents that are necessary or advisable in connection with such transactions. To the extent that any notices to, filings with or authorizations, consents or approvals from any Governmental Bodies are necessary or advisable in connection with such transactions, Matthews, the Acquiror, FDC, IEEC and the Shareholders shall give any such notices, make any such filings and use their respective best efforts to obtain any such authorizations, consents and approvals at Acquiror's cost.

3.02. Exclusivity. Until the Closing Date, neither FDC, IEEC nor the Shareholders shall (i) solicit, initiate or encourage the submission of any proposal or offer from any person relating to the acquisition of any capital stock or other voting securities or any of the assets of FDC or IEEC (including any acquisition structured as a merger, consolidation or share exchange) or to any other transaction which could result in control of FDC or IEEC by any person other than the Acquiror or (ii) participate in any discussions or negotiations regarding, furnish any information with respect to, assist or participate in, or facilitate in any other manner any effort or attempt by any person to do or seek any of the foregoing. The Shareholders shall not vote their Shares in favor of any such acquisition or transaction. The Shareholders shall notify the Acquiror immediately if any person makes any proposal, offer, inquiry or contact with respect to any such acquisition or transaction.

3.03. Operation and Preservation of Business. Until the Closing Date, neither FDC nor IEEC shall, without the prior consent of the Acquiror, engage in or permit any

practice, transaction or act (i) outside the Ordinary Course of Business or (ii) which, if it had otherwise been engaged in or permitted, would have rendered untrue in any material respect any of the representations and warranties of FDC, IEEC or the Shareholders contained in Article V. Without limiting the generality of the foregoing, neither FDC nor IEEC shall, without the prior consent of the Acquiror, declare, set aside or pay any dividend or make any distribution with respect to its capital stock or redeem, purchase or otherwise acquire any of its capital stock. FDC, IEEC or the Shareholders shall promptly notify the Acquiror of the occurrence of any matter or event which is material to the business, operations, properties, assets, or condition (financial or otherwise) of FDC or IEEC. Each of FDC and IEEC shall keep substantially intact its business and properties, including its present operations, physical facilities, working conditions, and relationships with lessors, licensors, suppliers, customers and employees.

### 3.04. Confidentiality

(a) Each of the Acquiror and Matthews shall treat as confidential any Confidential Information which it receives from or on behalf of FDC, IEEC or the Shareholders in the course of the investigation contemplated hereby and each agrees not to use any of the Confidential Information except in connection with the transactions contemplated by this Agreement.

(b) FDC, IEEC and the Shareholders shall treat as confidential all of the Confidential Information and agree not to use any of the Confidential Information except in connection with this Agreement.

(c) Each party shall deliver promptly to the other party or destroy, at the request of the other party, all tangible embodiments, including all copies thereof, of the Confidential Information which are within the possession or control of such party (which shall not include documents with respect to the transaction contemplated by this Agreement when the Closing occurs). If a party is requested or required (by oral question or request for information or documents in any legal proceeding, interrogatory, subpoena, civil investigative demand or similar process) to disclose any Confidential Information, such party shall notify the other party promptly of such request or requirement so that

the other party may seek an appropriate protective order or waive compliance with the provisions of this Section 3.04(c). If, in the absence of such a protective order or waiver, a party is, on the advice of counsel, compelled to disclose any Confidential Information to any tribunal or else be liable for contempt, such party may disclose such Confidential Information to such tribunal; provided, however, that the disclosing party shall use such party's best efforts to obtain, at the request of the other party, an order or other assurance that confidential treatment will be accorded to such portion of the Confidential Information required to be disclosed as the other party shall designate.

3.05. Disclosure Schedule. FDC, IEEC and the Shareholders shall promptly supplement the Disclosure Schedule if events occur prior to the Closing that would have been required to be disclosed had they existed at the time of executing this Agreement. The Disclosure Schedule, as supplemented prior to the Closing, will contain a true, correct and complete list and description of all items required to be set forth therein.

## ARTICLE IV Post-Closing Covenants

## After the Closing:

4.01. Further Assurances. FDC, IEEC and the Shareholders, on the one hand, and the Acquiror and Matthews, on the other hand, shall take all action and execute all documents, instruments or conveyances of any kind which may be requested by the Acquiror or FDC, IEEC and the Shareholders, respectively, in order to carry out any of the provisions hereof or consummate the transactions contemplated hereby, all at the sole cost and expense of the requesting party (unless the requesting party is or would be entitled to indemnification therefor under Article VI). FDC, IEEC and the Shareholders acknowledge and agree that, from and after the Closing, the Acquiror shall be entitled to possession of all documents, books, records (including Tax records), agreements and financial data of any sort relating to FDC and IEEC; provided, that FDC, IEEC or the Shareholders shall have access to such documents upon request for a period of seven years after the Closing.

4.02. Payment of Taxes. The Shareholders shall pay, or shall cause FDC to pay, all Taxes resulting from the transactions effected as a result of or prior to this Agreement, including without limitation all such Taxes associated with recapture of investment tax credits, depreciation recapture and other recaptured items. After making provisions for or satisfying FDC's liabilities as of the Closing Date, the Shareholders may prior to the closing date distribute as a dividend from FDC the amount of income for the 1996 tax year. that the Shareholders will be required to include in their 1996 tax returns by reason of FDC's S-corporation status. The Shareholders shall cause IEEC to include in all appropriate Tax Returns, the operations of IEEC to the Closing Date and shall cause such Tax Returns to be prepared at the Acquiror's expense and shall sign and file such final Tax Returns. The Acquiror shall pay all Taxes due as shown on such Tax Returns.

4.03. Matthews Stock. Each Shareholder agrees that such Shareholder will not, directly or indirectly, offer for sale, sell, transfer or otherwise dispose of, any shares of Matthews Stock except in accordance with (I) the provisions of the Securities Act and the General Rules and Regulations thereunder, including, but not limited to, Rule 144, (ii) any applicable requirements of the Securities Exchange Act of 1934, as amended, and (iii) any other similar Federal or state requirement then in force. Each Shareholder

agrees to and does hereby indemnify and hold harmless Matthews and the Acquiror, the officers and directors of Matthews and the Acquiror and each other person who controls Matthews and the Acquiror within the meaning of the Securities Act against any and all losses, claims, damages, liabilities and expenses which Matthews, the Acquiror or any such officers, directors or other persons may incur arising out of any offer, sale, transfer or disposition of shares of Matthews Stock by or on behalf of such Shareholder in violation of this Section 4.03 or any other provision of this Agreement.

4.04. Legends. Each Shareholder agrees that the certificates representing Matthews Stock to be received in connection with the Merger may bear the following legend:

"These securities have not been registered under the Securities Act of 1933 or any state securities law. They may not be sold, offered for sale, pledged, hypothecated or otherwise disposed of in the absence of a registration statement in effect with respect to the securities under the Securities Act and each applicable state securities law or an opinion of counsel satisfactory to Matthews International Corporation that such registration is not required."

4.05. Employee Benefits. The parties agree to undertake an examination of the respective benefit plans offered by Matthews and IEEC (after the merger, the Acquiror) and to determine by mutual agreement whether or not IEEC's employees will receive Matthews' benefit plans or whether such employees will retain the benefit plans currently offered by IEEC. For a period of five (5)years after the Closing Date, neither the Acquiror nor Matthews will change the benefits currently offered to IEEC employees without prior written consent of each of the Shareholders.

## ARTICLE V REPRESENTATIONS AND WARRANTIES

5.01. Representations and Warranties of FDC, IEEC and the Shareholders. FDC, IEEC and the Shareholders represent and warrant to the Acquiror that the statements contained in this Section 5.01 are correct and complete in all material respects as of the date of this Agreement and will be correct and complete in all material respects as of the Closing Date (as though made then and as though the Closing Date were substituted for the date of this Agreement throughout this Section 5.01), except in each case as set forth in the Disclosure Schedule. Nothing in the Disclosure Schedule shall be deemed adequate to disclose an exception to a representation or warranty made herein unless the Disclosure Schedule identifies the exception with particularity.

(a) Organization, Qualification and Corporate Power. Each of FDC and IEEC is a corporation duly organized, validly existing and in good standing under the

laws of the State of Florida. Each of FDC and IEEC has all corporate power and authority and all licenses, permits and authorizations (including qualification in each other jurisdiction where such qualification is required) necessary to own, lease and use its properties and to carry on the businesses in which it is engaged as and where such properties are now owned, leased and used by it and such practice and businesses are now carried on. Section 5.01(a) of the Disclosure Schedule sets forth a true and complete list of each jurisdiction in which each of FDC and IEEC is qualified to do business as a foreign corporation. Section 5.01 (a) of the Disclosure Schedule also lists the directors and officers of each of FDC and IEEC. Each of FDC and IEEC has delivered to the Acquiror correct and complete copies of its articles of incorporation and bylaws as in effect on the date of this Agreement. The minute books (containing the records of all meetings of the Shareholders, the Board of Directors and any committees of the Board of Directors) and the stock certificate books of each of FDC and IEEC are correct and complete. Neither FDC or IEEC is in default under or in violation of any provision of its articles of incorporation or bylaws.

(b) Authorization of Transaction; Enforceability.

(i) Each of FDC and IEEC has full corporate power and authority and has taken all necessary action to execute, deliver and consummate this Agreement and to perform its obligations hereunder. Each of the Shareholders has the legal capacity and has taken all necessary action to execute, deliver and consummate this Agreement, the Employment Agreements and the Registration Rights Agreement and to perform the obligations of the Shareholders hereunder and thereunder.

(ii) This Agreement constitutes the valid and legally binding obligations of each of the Shareholders and each of FDC and IEEC, enforceable in accordance with its terms and conditions, and when executed and delivered by the parties thereto, the Employment Agreements and the Registration Rights Agreement will constitute the valid and legally binding obligation of the Shareholders, enforceable in accordance with its terms and conditions, except in each case as the enforceability thereof may be limited by bankruptcy, insolvency or other laws of general application relating to or affecting the enforcement of creditors' rights or by general principles of equity limiting the availability of equitable remedies. (iii) Neither FDC, IEEC nor the Shareholders are required to give any notice to, make any filing with, or obtain any authorization, consent or approval of, any Governmental Body in order to consummate the transactions contemplated by this Agreement, the Employment Agreements and the Registration Rights Agreement.

(c) No Contravention. Neither the execution and delivery of this Agreement, the Employment Agreements and the Registration Rights Agreement, nor the consummation of the transactions contemplated hereby or thereby, does or will
(i) violate any Law or the articles of incorporation or bylaws of either FDC or IEEC, or (ii)

except as set forth in Section 5.01 (c) of the Disclosure Schedule, conflict with, result in a material breach of, constitute a material default under, result in the acceleration of, create in any party the right to accelerate, terminate, modify or cancel, or require any notice under, any material agreement, contract, lease, license, instrument or other arrangement to which FDC, IEEC or the Shareholders are a party or by which FDC, IEEC or the Shareholders are subject (or result in the imposition of any Lien upon any of the assets of FDC, IEEC or the Shareholders).

(d) Capitalization of Ownership of Shares.

(i) The entire authorized capital stock of FDC consists of 7,500 shares of common stock, par value \$1 per share, of which 200 shares are issued and outstanding. The entire authorized capital stock of IEEC consists of 1000 shares of common stock, par value \$1 per share, of which 1,000 shares are issued and outstanding. There are no outstanding or authorized options, warrants, purchase rights, subscription rights, conversion rights, exchange rights or other contracts or commitments that could require either FDC or IEEC to issue, sell or otherwise cause to become outstanding any of its capital stock. There are no voting trusts, proxies or other agreements or understandings with respect to the voting of the capital stock of either FDC or IEEC.

(ii) On the date hereof and on the Closing Date, the Shareholders hold and will hold of record and own and will own beneficially all the Shares of FDC and IEEC, free and clear of any restrictions on transfer, Liens, options, warrants, purchase rights, contracts, commitments, equities, claims and demands except for this Agreement, and except for any restrictions under the Securities Act and state corporate and securities Laws. No Shareholder is a party to any option, warrant, purchase right or other contract or commitment that could require the Shareholder to sell, transfer or otherwise dispose of any capital stock or other securities of FDC or IEEC.

(e) Subsidiaries. Neither FDC or IEEC owns, directly or indirectly, any equity or debt securities issued by any other corporation, or any interest in any partnership, joint venture or other business enterprise.

(f) Financial Statements; Liabilities; Guarantees; No Material Adverse Change

(i) IEEC has delivered to the Acquiror its balance sheets, statements of income, statements of changes in shareholders' equity and statements of cash flow as of December 31, 1994 and 1993 and for the fiscal years then ended, each as reviewed by independent accountants, together with the reports thereon of such independent accountants. IEEC has also delivered to the Acquiror such Financial

Statements for the fiscal year ended December 31, 1995, as audited by BDO Seidman, and as attached hereto as Exhibit C. FDC has delivered to the Acquiror copies of its Federal income tax returns for the years ended December 31, 1993, 1994 and 1995. The Financial Statements for all of such years (including the notes thereto) present fairly in all material respects, (provided, however, that Financial Statements that are not audited may exclude footnotes that may be otherwise required by generally accepted accounting principles), the financial condition of FDC and IEEC as of the end of each such fiscal year and the results of their respective operations and cash flows for the fiscal year then ended. Such Financial Statements are correct and complete in all material respects and are consistent with the books and records of each of FDC and IEEC (which books and records are correct and complete). (ii) Except for Liabilities arising under this Agreement, there are no material aggregate material Liabilities of any nature of FDC and IEEC, whether or not accrued or contingent and whether or not determined or determinable, except as disclosed in the Financial Statements or in the Disclosure Schedules, and to the knowledge of FDC, IEEC and the Shareholders, there is no Basis for any present or future action, suit, proceeding, hearing, investigation, charge, complaint, claim or demand against FDC and IEEC which might give rise to any Liability, other than Liabilities fully set forth or reflected in the Financial Statements or Liabilities incurred after December 31, 1995 in the Ordinary Course of Business and whose existence would not cause a breach of any provision of this Agreement and would not cause any of the representations and warranties in this Section 5.01 to be inaccurate or incomplete. Except as disclosed in the Financial Statements, neither FDC nor IEEC is indebted to the Shareholders or any officer, director or employee of FDC or IEEC or their respective families, except for compensation and accrued vacation pay reflected on the books and accounts of FDC or IEEC, as the case may be.

(iii) Neither FDC nor IEEC is a guarantor or otherwise liable for any Liability or obligation (including indebtedness) of any other person.

(iv) Since December 31, 1995, neither FDC nor IEEC has suffered any change in its condition (financial or otherwise) or in its assets, operations, properties, liabilities (absolute or contingent), or business, except changes which are lawful and in the Ordinary Course of Business, and which individually or in the aggregate, have not materially and adversely affected its condition (financial or otherwise), assets, operations, properties, or business.

(g) Real Property; Leases; Assets; Liens; Condition of Properties.

(i) All of the interests of FDC and IEEC in real property owned in fee ("Real Property") are listed and described in Section 5.01 (g)(1) of the Disclosure Schedule. Each of FDC and IEEC has good and marketable title to Real Property, free and clear, in each instance, of any and all Liens, which are material in character,

amount or extent or which materially detract from the value or materially interfere with the use, as currently utilized of the properties subject thereto or affected thereby, or otherwise materially impair the business operations being conducted thereon. The Real Property includes all material easements and rights-of-way necessary for present access to and use, as currently utilized, of the Real Property. The Real Property conforms to all applicable zoning laws and regulations and no written notice of violation of any such laws or regulations relating to any of the material Real Property has been received by either FDC or IEEC. To the knowledge of FDC and IEEC, no condemnation proceedings are proposed, threatened or pending which would materially affect the Real Property.

(ii) All of the leases under which FDC or IEEC leases real or personal property are listed and described in Section 5.01(g)(ii) of the Disclosure Schedule. All such leases are valid and subsisting and no default exists under any thereof. FDC and IEEC own all other property and assets reflected on their Most Recent Balance Sheet or presently used by them or located on their premises, except personal property transferred, conveyed or otherwise disposed of since December 31, 1995 all of which was transferred, conveyed or otherwise disposed of in the Ordinary Course of Business. None of the personal property owned by FDC and IEEC, and none of the leasehold interests of FDC and IEEC in any real or personal property leased by FDC and IEEC, is subject to any Lien or restriction upon its use or disposition except as otherwise stated in Section 5.01 (g)(ii) of the Disclosure Schedule. There are no outstanding options or rights in any person to acquire any of such leased or other property or assets or any interest therein except as otherwise stated in Section 5.01(g)(ii) of the Disclosure Schedule.

(iii) Each of FDC and IEEC owns or leases all buildings, machinery and other equipment, supplies and other tangible assets necessary for the conduct of its business as presently conducted. Neither FDC nor IEEC occupies or uses any property of others except under valid and enforceable leases, contracts or other arrangements. To the knowledge of FDC, IEEC and the Shareholders, all such tangible assets are free from material defects and all such buildings are in a good state of repair, have been regularly maintained, substantially conform with all applicable ordinances, regulations and zoning or other Laws (including Environmental, Health and Safety Laws) and do not encroach on property of others. All such equipment is in good working order and substantially conforms with all applicable ordinances, regulations and zoning or other Laws (including Environmental, Health and Safety Laws). There is no pending or to the knowledge of FDC, IEEC and the Shareholders threatened change of any such ordinance, regulation or zoning or other Law which might materially and adversely affect the business, operations or property of either FDC or IEEC, and there is no pending or to the knowledge of FDC, IEEC and the Shareholders threatened condemnation of any such property.

(h) Accounts Receivable, Reimbursement Claims. All notes and accounts receivable of FDC and IEEC reflected on their respective books and records are valid obligations subject to no setoffs or counter-claims known to IEEC, are current except as set forth in the Accounts Receivable Aging Analysis set forth in Section 5.01(h) of the Disclosure Schedule and are collectible, and will be collected in accordance with their terms at their recorded amounts, subject only to the reserve for bad debts set forth in their respective Most Recent Balance Sheet, as adjusted for the passage of time through the Closing Date in accordance with the past custom and practice of FDC and IEEC.

(i) Contracts. Except only for this Agreement, the Employment Agreements and the contracts, agreements and commitments which are identified in Section 5.01
(i) of the Disclosure Schedule by title, date, parties and nature of contract, with specific reference to the relevant clause of this Section 5.01 (i), neither FDC, IEEC nor the Shareholders are a party to or subject to any of the following, whether written or oral:

(i) Any management or employment contract or contract for personal services with any employee, agent or consultant which will not be terminable at will without penalty by the Acquiror immediately after the Closing;

(ii) Any contract or arrangement with respect to the purchase of equipment or supplies, or professional or non-professional services, or insurance, which (A) involves expenditures after December 31, 1995 of more than \$50,000 in the aggregate or (B) otherwise is material;

(iii) Any contract or agreement containing covenants by FDC, IEEC or the Shareholders not to compete in any line of business with any person, or not to disclose to any person any information, documents, plans or any other intellectual property, tangible or intangible;

(iv) Any contract or agreement with any labor union;

(v) Any contract, commitment or agreement for the sale of products or services for \$50,000 or more in the aggregate to a single customer or a group of related customers;

(vi) Any contract or agreement under which there is outstanding indebtedness for borrowed money or a right to borrow money;

(vii) Any contract, agreement, arrangement or understanding not in the Ordinary Course of Business, or in violation of any Law, or upon which any material part of the-business of FDC or IEEC is dependent, or which could materially

and adversely affect the business, assets, properties, condition (financial or otherwise) or prospects of FDC or IEEC; or

(viii) Any contract which requires that any consent be obtained or notice given in order for such contract to be assigned or transferred to the Acquiror.

The Shareholders have delivered to the Acquiror a correct and complete copy of each written agreement (as amended to date) listed in Section 5.01(1) of the Disclosure Schedule and a written summary setting forth the terms and conditions of each oral agreement, if any, referred to in Section 5.01(1) of the Disclosure Schedule. With respect to each such agreement: (A) the agreement is legal, valid, binding, enforceable and in full force and effect; (B) Except as set forth in Section 5.01 (i) of the Disclosure Schedule, the agreement is assignable to the Acquiror and will continue to be legal, valid, binding, enforceable and in full force and effect on identical terms immediately after the Closing Date; (C) to the knowledge of FDC and IEEC no party thereto is in breach or default, and no event has occurred which with notice or lapse of time would constitute a breach or default, or permit termination, modification or acceleration, under the agreement; and (D) no party thereto has repudiated any provision of the agreement.

(j) Intellectual Property.

(i) FDC and IEEC own or have the right to use pursuant to license, sublicense, agreement or permission all Intellectual Property necessary for the conduct of their business as presently conducted. Each such item of Intellectual Property so owned or used immediately prior to the Closing will be owned or available for use by the Acquiror on identical terms and conditions immediately after the Closing. FDC and IEEC have taken all necessary action to maintain and protect each such item of Intellectual Property that either so owns or uses.

(ii) Neither FDC, IEEC nor the Shareholders, to their knowledge, have interfered with, infringed upon, misappropriated or otherwise come into conflict with any rights of third parties with respect to Intellectual Property necessary for the conduct of the business of FDC and IEEC, or, within the past three years, have received any charge, complaint, claim, demand or notice alleging any such interference, infringement, misappropriation or violation (including any claim that FDC, IEEC or the Shareholders must license or refrain from using any Intellectual Property rights of any third party). To the Knowledge of FDC, IEEC and the Shareholders, no third party has interfered with, infringed upon, misappropriated or otherwise come into conflict with any Intellectual Property rights of FDC and IEEC necessary or for the conduct of their businesses.

(iii) Section 5.010(iii) of the Disclosure Schedule (A) identifies each patent, trademark or service mark registration which has been issued to FDC,

IEEC or the Shareholders with respect to any such Intellectual Property necessary for the conduct of the business of FDC or IEEC, identifies each pending patent, trademark or service mark application which FDC, IEEC or the Shareholders have made with respect to any such Intellectual Property, and identifies each license, agreement or other permission which FDC, IEEC or the Shareholders have granted to any third party with respect to any Intellectual Property; (B) identifies each trade name or unregistered trademark used by FDC or IEEC in connection with their respective businesses; and identifies each item of Intellectual Property that any third party owns and that FDC, IEEC or the Shareholders use in the conduct of the respective businesses of FDC or IEEC pursuant to license, sublicense, agreement or permission.

(iv) The Shareholders do not own or have the right to use any Intellectual Property necessary for the conduct of the respective businesses of FDC or IEEC which may not be used after the Closing by the Acquiror without material restriction or fee.

(k) Litigation; Liabilities; Legal Compliance.

(i) Section 5.01 (k)(1) of the Disclosure Schedule sets forth each instance in which FDC or IEEC, or any of their employees, (A) is, or has been within the past three years, subject to any outstanding injunction, judgment, order, decree, ruling or charge or (B) is, or has been within the past three years, a party or, to the Knowledge of the Shareholders and the directors and officers (and employees with responsibility for litigation matters) of FDC or IEEC, is, or has been within the past three years, threatened to be made a party, to any action, suit, proceeding, hearing or investigation of, in or before any court or other Governmental Body or before any arbitrator in any United States or foreign jurisdiction. None of the actions, suits, proceedings, hearings or investigations required to be set forth in Section 5.01 (k)(i) of the Disclosure Schedule will result in any adverse change in the business, condition (financial or otherwise), operations, results of operations or future prospects of FDC or IEEC.

(ii) Neither FDC or IEEC nor any director, officer or employee of FDC or IEEC, nor the Shareholders have or on the Closing Date will have any Liability not fully covered by insurance (and to the knowledge of FDC, IEEC and the Shareholders, there is no Basis for any present or future action, suit, proceeding, hearing, investigation, charge, complaint, claim or demand against any of them giving rise to any Liability not fully covered by insurance) arising out of any injury to individuals or property as a result of any action or failure to act by FDC or IEEC, or any such director, officer or employee, or the Shareholders.

(iii) FDC, IEEC and the Shareholders have complied in all material respects with material Laws applicable to the respective businesses of FDC and IEEC.

(1) Insurance. Section 5.01(1) of the Disclosure Schedule sets forth the following information with respect to each insurance policy (including policies providing professional liability, property, casualty, other liability and workers' compensation coverage and bond and surety arrangements) relating the businesses of FDC and IEEC to which FDC or IEEC, or any of their directors, officers or employees, or any Shareholder has been a party, a named insured or otherwise the beneficiary of coverage at any time within the past six years:

(i) the name of the insurer, the name of the policyholder, the policy number and the period of coverage;

(ii) the scope (including whether the coverage was on a claims made, occurrence or other basis) and amount of coverage (including a description of how deductibles, co-payment obligations and ceilings are calculated and operate); and

(iii) a description of any retroactive premium adjustment or other loss-sharing arrangements.

Except as set forth in Section 5.01(1) of the Disclosure Schedule, with respect to each such insurance policy: (A) the policy is legal, valid, binding, enforceable and in full force and effect; (B) the policy will continue to be legal, valid, binding, enforceable and in full force and effect on identical terms immediately after the Closing, unless, upon mutual agreement of FDC and IEEC and the Acquiror, coverage is replaced; (C) neither FDC, IEEC nor the Shareholders nor any other party to the policy is in breach or default (including with respect to the payment of premiums or the giving of notices), and no event has occurred which, with notice or the lapse of time, would constitute such a breach or default, or permit termination, modification or acceleration, under the policy; and (D) no party to the policy has repudiated any provision thereof.

(m) Tax Matters. FDC and IEEC have filed all Tax Returns that each was required to file. All such Tax Returns are correct and complete in all material respects. All Taxes due and payable by FDC or IEEC as shown on any Tax Return have been paid. With respect to any taxable period for which the statute of limitations has not expired, neither FDC nor IEEC has waived any statute of limitations in respect of Taxes or agreed to any extension of time with respect to a Tax assessment or deficiency. FDC and IEEC have withheld and paid all Taxes required to have been withheld and paid in connection with amounts paid or owing to any employee. Except as disclosed in Section 5.01(m) of the Disclosure Schedules, there is no dispute or claim concerning any Liability of FDC or IEEC for Taxes either (A) claimed or raised by any authority in writing or (B) as to which either FDC or IEEC otherwise has Knowledge.

(n) Employees. Section 5.01(n) of the Disclosure Schedule identifies:

(a) each management or employment contract or contract for personal services with any officer, consultant, salesman or other non-union employee or agent of FDC or IEEC, and

(b) each officer, consultant, salesman or other non-union employee or agent of FDC or IEEC receiving as at the date hereof compensation (including bonuses and commissions, if any) at an annual rate of \$30,000 or more, specifying the rate of such compensation (including such bonuses and commissions) and the positions held by such persons.

FDC and IEEC have heretofore made available to the Acquiror true and complete copies of each of the contracts referred to above as in effect on the date hereof. There are no material disputes presently subject to any grievance procedure, arbitration or litigation or threatened litigation under any such contract nor is there any material default, or any event which with notice or the lapse of time or both will become a material default, under any such contract, by FDC or IEEC or, to the knowledge of FDC, IEEC or the Shareholders, any other party thereto. Neither FDC nor IEEC is a party to or bound by any collective bargaining agreement. Neither FDC nor IEEC has committed any unfair labor practice. Neither the Shareholders nor any of the directors and officers (and employees with responsibility for employment matters) of FDC or IEEC has

any Knowledge of any organizational effort presently being made or threatened by or on behalf of any labor union with respect to employees of FDC or IEEC.

(o) Employee Benefits.

(i) Section 5.01(0)(1) of the Disclosure Schedule lists each Employee Benefit Plan that FDC or IEEC maintains or to which FDC or IEEC contributes.

(A) Each such Employee Benefit Plan (and each related trust, insurance contract or fund) complies in form and operation in all respects with the applicable requirements of ERISA, the Code and other applicable Laws.

(B) All required reports and descriptions (including Form 5500 Annual Reports, Summary Annual Reports, PBGC-I's and Summary Plan Descriptions) have been filed or distributed appropriately with respect to each such Employee Benefit Plan. The requirements of Part 6 of Subtitle B of Title I of ERISA and of Code Section 4980B have been met with respect to each such Employee Benefit Plan which is an Employee Welfare Benefit Plan.

(C) All contributions (including all employer contributions and employee salary reduction contributions) which are due have been paid to each such Employee Benefit Plan which is an Employee Pension Benefit Plan, and all contributions for any period

ending on or before the Closing Date which are not yet due have been or will be paid to each such Employee Pension Benefit Plan before the Closing Date. All premiums or other payments for all periods ending on or before the Closing Date have been paid, or will be paid before the Closing Date, with respect to each such Employee Benefit Plan which is an Employee Welfare Benefit Plan.

(D) Each such Employee Benefit Plan, which is an Employee Pension Benefit Plan, meets the requirements of a "qualified plan" under Code Section 401(a) and has received a favorable determination letter from the Internal Revenue Service.

(E) The market value of assets under each such Employee Benefit Plan which is an Employee Pension Benefit Plan (other than any Multi employer Plan) equals or exceeds the present value of all vested and nonvested Liabilities thereunder, determined in accordance with PBGC methods, factors and assumptions applicable to an Employee Pension Benefit Plan terminating on the date for determination.

(F) FDC, IEEC and the Shareholders have delivered to the Acquiror correct and complete copies of the plan documents and summary plan descriptions, the most recent determination letter received from the Internal Revenue Service, the most recent Form 5500 Annual Report and all related trust agreements, insurance contracts and other funding agreements which implement each such Employee Benefit Plan.

(ii) With respect to each Employee Benefit Plan that FDC or IEEC maintains or ever has maintained or to which he contributes, ever has contributed or ever has been required to contribute:

(A) No such Employee Benefit Plan which is an Employee Pension Benefit Plan (other than any Multi employer Plan) has been completely or partially terminated or been the subject of a Reportable Event as to which notices would be required to be filed with the PBGC. No proceeding by the PBGC to terminate any such Employee Pension Benefit Plan (other than any Multi employer Plan) has been instituted or, to the Knowledge of any of the Shareholders and the directors and officers (and the employees with responsibility for employee benefits matters) of FDC or IEEC, threatened.

(B) There have been no Prohibited Transactions with respect to any such Employee Benefit Plan. No Fiduciary has any Liability for breach of fiduciary duty or any other failure to act or comply in connection with the administration or investment of the assets of any such Employee Benefit Plan. No action, suit, proceeding, hearing or investigation with respect to the administration or the investment of the assets of any such Employee Benefit Plan (other than routine claims for benefits) is pending or, to the Knowledge of any of the Shareholders and the directors and officers (and employees with responsibility for employee benefits matters) of FDC or IEEC, threatened. Neither the Shareholders nor the directors and officers (and employees with responsibility for employee benefits matters) of FDC or IEEC have any Knowledge of any Basis for any such action, suit, proceeding, hearing or investigation.

(C) Neither FDC or IEEC have incurred, and none of the Shareholders or the directors and officers (and employees with responsibility for employee benefits matters) of FDC or IEEC have any reason to expect that FDC or IEEC will incur, any Liability to the PBGC (other than PBGC premium payments) or otherwise under Title IV of ERISA (including any withdrawal Liability) or under the Code with respect to any such Employee Benefit Plan which is an Employee Pension Benefit Plan.

(iii) Neither FDC nor IEEC have contributed to or been required to contribute to any Multi employer Plan, and neither FDC nor IEEC has Liability (including withdrawal Liability) under any Multi employer Plan.

(iv) Neither FDC nor IEEC maintained or contributed to or been required to contribute to any Employee Welfare Benefit Plan providing medical, health or life insurance or other welfare-type benefits for current or future retired or terminated employees, their spouses or their dependents (other than in accordance with Code Section 4980B).

(p) Employee Severance Claims. Provided that it complies with the Warn Act, the Acquiror shall not have any Liability in respect of severance or separation pay to Persons employed by FDC or IEEC on the Closing Date as a result of the consummation of this Agreement and the transactions contemplated hereby.

(q) Environment, Health and Safety.

(i) FDC and IEEC, to their knowledge, have been and are in compliance in all material respects with all Environmental, Health and Safety Laws, and no action, suit, proceeding, hearing, investigation, charge, complaint, claim, demand or notice has been filed or commenced against FDC or IEEC alleging any failure so to comply. Without limiting the generality of the preceding sentence, FDC and IEEC have obtained, and have been and are in compliance in all material respects with all of the terms and conditions of, all material permits, licenses and other authorizations which are required under, and has complied with all other limitations, restrictions, conditions, standards, prohibitions, requirements, obligations, schedules and timetables which are contained in, all material Environmental, Health and Safety Laws applicable to them. No such permits, licenses or other authorizations have expired, have been administratively extended or will lapse or become void as a result of the transactions contemplated by this Agreement and the Employment Agreements.

(ii) Neither FDC nor IEEC has material Liability (and neither FDC nor IEEC has handled or disposed of any substance, arranged for the disposal of any substance, exposed any employee or other individual to any substance or condition,

or owned or operated any property or facility, in any manner that could form the Basis for any present or future action, suit, proceeding, hearing, investigation, charge, complaint, claim or demand against them giving rise to any Liability) for response or remediation costs, fines or penalties at any site, location or body of water (surface or subsurface), for any illness of or personal injury to any employee or other individual, or for any other reason, under any Environmental, Health or Safety Law.

(iii) All properties and equipment used in the businesses of FDC and IEEC are and have been free of asbestos, PCB's, underground storage tanks, methylene chloride, trichloroethylene, 1,2-trans-dichloroethylene, dioxins, dibenzofurans and Extremely Hazardous Substances.

(r) Governmental Licenses and Permits. All licenses and permits (collectively, the "Permits") material to and necessary in the conduct of the business or operation of FDC and IEEC, or to the ownership or use of their Assets, have been obtained and are in full force and effect, and no material violation of any such Permit has occurred and no proceeding is pending or to the knowledge of FDC and IEEC threatened to revoke or limit any such Permit. Section 5.01(r) of the Disclosure Schedule identifies all Permits relating to the business, assets or operations of FDC and IEEC. FDC, IEEC and the Shareholders have delivered to the Acquiror true and complete copies of all Permits as in effect on the Closing Date.

(s) Powers of Attorney; Bank Accounts. There are no outstanding powers of attorney executed on behalf of FDC and IEEC. Section 5.01(s) of the Disclosure Schedule identifies all bank accounts owned by FDC or IEEC or to which FDC or IEEC has access.

(t) Certain Business Relationships with FDC or IEEC. Neither the Shareholders nor their Affiliates have been involved in any business arrangement or relationship with FDC or IEEC, including the receipt of any loan or advance from FDC or IEEC, and neither the Shareholders nor their Affiliates owns any interest in any tangible or intangible asset which is used in the business of FDC or IEEC, except in each case as set forth in Section 5.01(t) of the Disclosure Schedule.

(u) Customers and Suppliers. Section 5.01(u) of the Disclosure Schedule identifies each of the customers and suppliers of FDC or IEEC whose purchases from or sales to FDC or IEEC constituted five (5) percent or more of the combined net sales revenues or net purchases, respectively, of FDC or IEEC during the year ended December 31, 1995, showing, with respect to each, the name and address, dollar volume and nature of the relationship (including the principal categories of products leased or sold).

(v) Inventory. The inventory of FDC and IEEC is in all material respects of a quality and quantity usable or salable in the ordinary course of business

as heretofore conducted and all such inventory reflected on the Most Recent Balance Sheet of FDC and IEEC is reflected with adequate provisions for excess inventory, slow-moving inventory and inventory obsolescence and shrinkage.

(w) Warranty Expense. The aggregate expense incurred by FDC or IEEC to satisfy warranty claims has not exceeded \$100,000 in any of their past 5 fiscal years, Section 5.01(w) of the Disclosure Schedule sets forth a description of all warranties provided by FDC or IEEC.

(x) Brokers' Fees. Neither FDC, IEEC nor the Shareholders have any Liability or obligation to pay any fees or commissions to any broker, finder or agent with respect to the transactions contemplated by this Agreement or any Employment Agreement.

(y) Full Disclosure. The representations and warranties contained in this Section 5.01 do not contain any untrue statement of a material fact or omit to state any material fact necessary in order to make the statements and information contained or referred to in this Section 5.01 not misleading.

(z) Acquisition Entirely for Own Account. The Matthews Stock to be received in connection with the Merger by each Shareholder will be acquired for investment for such Shareholder's own account, not as a nominee or agent, and not with a view to the resale or distribution of any part thereof, and neither Shareholder has the present intention of selling, granting any participation in, or otherwise distributing the same. Each Shareholder further represents and warrants that such Shareholder does not have any contract, undertaking, agreement or arrangement with any Person to sell, transfer or grant participation to such Person or to any third Person, with respect to any of the Matthews Stock.

(aa) Disclosure of Information. Each Shareholder has received all the information it considers necessary or appropriate for deciding whether to vote its Shares for approval of this Agreement and the Plan of Merger and receive shares of Matthews Stock in connection with the Merger. Each Shareholder acknowledges receipt of (a) (i) the Form 10-K Annual Reports of Matthews for the fiscal years ended September 30, 1993-1995, (ii) the Annual Reports to Shareholders of Matthews for the years ended September 30, 1993-1995, and (iii) the Proxy Statements of Matthews for each annual or special meeting of Matthews shareholders between September 30, 1993 and the date hereof, (b) the Form 10-Q Quarterly Reports of Matthews for the first fiscal quarter of the fiscal year beginning October 1, 1995 and (c) each Form 8-K Current Report of Matthews filed since October 1, 1995. (All reports and statements referred to in this Section 5.01 (aa) are referred to collectively as "Matthews' Exchange Act Filings"). Each Shareholder has had an opportunity to ask questions and receive answers from Matthews regarding the business and affairs of Matthews and the terms and provisions of the Matthews Stock.

(bb) Restricted Securities.

(i) Each Shareholder understands that the shares of Matthews Stock it is receiving in connection with the Merger are characterized as "restricted securities" under the Securities Act and the rules and regulations thereunder inasmuch as they will not have been registered under the Securities Act and they are being acquired from Matthews in a transaction not involving a public offering and that under the Securities Act and rules and regulations such securities may not be sold, offered for sale, pledged, hypothecated or otherwise disposed of without registration under the Securities Act, or unless an exemption from registration is available. In this connection, each Shareholder represents that it is familiar with Rule 144 under the Securities Act, as presently in effect, and understands the resale limitations imposed thereby and by the Securities Act.

(ii) Each Shareholder understands that the shares of Matthews Stock it is receiving in connection with the Merger are not being registered under the securities or "blue sky" laws of any jurisdiction and may not be sold, offered for sale, pledged, hypothecated or otherwise disposed of without registration under any such laws as may be applicable or an exemption from the registration requirements thereof.

(cc) Accredited Investor. Each shareholder is a natural person (I) whose individual net worth, or joint net worth with his spouse, at the date hereof and at the Effective Date exceeds \$1,000,000 or (ii) who had an individual income in excess of \$200,000 in each of the two most recent years and who reasonably expects an income in excess of \$200,000 in the current year.

(dd) Blue Sky.

(i) Neither FDC or IEEC has (A) sold securities to more than ten persons from the date of initial capitalization or (B) made offers to sell securities to more than 90 persons in the Commonwealth of Pennsylvania in a consecutive 12-month period.

(ii) Neither FDC nor IEEC nor any promoter, officer or director of FDC or IEEC is subject to the disqualifications set forth in 204.010(b) of the Blue Sky regulations Of the Commonwealth of Pennsylvania.

5.02. Representations and Warranties of the Acquiror and Matthews. Each of the Acquiror and Matthews represents and warrants to FDC, IEEC and the Shareholders that the statements contained in this Section 5.02 are correct and complete as of the date of this Agreement and will be correct and complete as of the Closing Date (as though made then and as though the Closing Date was substituted for the date of this Agreement throughout this Section 5.02).

(a) Organization and Corporate Power. Each of the Acquiror and Matthews is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware. Each of the Acquiror and Matthews has all power and authority (corporate and other) and all licenses, permits and authorizations necessary to own, lease and use its properties and to carry on the businesses in which it is engaged as and where such properties are now owned, leased and used by it and such businesses are now being conducted.

(b) Authorization of Transaction.

(i) Each of the Acquiror and Matthews has full corporate power and authority and has taken all necessary action to execute, deliver and consummate this Agreement, the Employment Agreement and the Registration Rights Agreement and to perform their respective obligations hereunder and thereunder.

(ii) This Agreement constitutes the valid and legally binding obligations of the Acquiror and Matthews, enforceable in accordance with its terms and conditions, and when executed and delivered by the parties thereto, the Employment Agreements and the Registration Rights Agreement will constitute the valid and legally binding obligations of the Acquiror and Matthews, enforceable in accordance with their respective terms and conditions, except in each case as the enforceability thereof may be limited by bankruptcy, insolvency or other laws of general application relating to or affecting the enforcement of creditors' rights or by general principles of equity limiting the availability of equitable remedies. (iii) Neither the Acquiror nor Matthews is required to give any notice to, make any filing with, or obtain any authorization, consent or approval of, any Governmental Body in order to consummate the transactions contemplated by this Agreement and the Employment Agreements.

(c) No Contravention. Neither the execution and delivery of this Agreement, the Employment Agreements and the Registration Rights Agreement nor the consummation of the transactions contemplated hereby or thereby, does or will (i) violate any Law applicable to or the articles of incorporation or bylaws of the Acquiror or Matthews or (ii) conflict with, result in a breach of, constitute a default under, result in the acceleration of, create in any party the right to accelerate, terminate, modify or cancel, or require any notice under, any agreement, contract, lease, license, instrument or other arrangement to which the Acquiror or Matthews is a party or by which the Acquiror or Matthews is subject (or result in the imposition of any Lien upon any of the assets of the Acquiror or Matthews).

(d) Brokers' Fees. Neither the Acquiror nor Matthews has any Liability or obligation to pay any fees or commissions to any broker, finder or agent with respect

to the transactions contemplated by this Agreement, any Employment Agreement or the Registration Rights Agreement.

(e) Matthews' Exchange Act Filings.

(i) As of the respective dates of each of Matthews' Exchange Act Filings, to the knowledge of Matthews each such filing complied in all material respects with the applicable requirements of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), and to the knowledge of Matthews did not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading. The audited consolidated financial statements and unaudited consolidated interim financial statements of Matthews included in Matthews' Exchange Act Filings have been prepared in accordance with generally accepted accounting principals applied on a consistent basis during the periods involved, and present fairly in all material respects the financial condition of Matthews as of the dates thereof and the results of Matthews' operations and cash flows for the periods then ended (subject, in the case of unaudited interim financial statements, to year-end adjustments).

(ii) Except for Liabilities arising under this Agreement, to the knowledge of Matthews there are no material Liabilities of any nature of the Acquiror or Matthews, whether or not accrued or contingent and whether or not determined or determinable, and, to the knowledge of the Acquiror and Matthews, there is no Basis for any present or future action, suit, proceeding, hearing, investigation, charge, complaint, claim or demand against the Acquiror or Matthews which might give rise to any material Liability, other than Liabilities fully set forth or reflected in Matthews' Exchange Act Filings or Liabilities incurred after December 31, 1995 in the ordinary course of business of Matthews consistent with its past custom and practice and whose existence would not cause a breach of any provision of this Agreement and would not cause any of the representations and warranties in this Section 5.02 to be inaccurate or incomplete.

(iii) Since December 31, 1995, neither the Acquiror nor Matthews has suffered any change in its condition (financial or otherwise) or in its assets, operations, properties, liabilities (absolute or contingent), or business, except changes which are lawful and in the Acquiror's or Matthews' ordinary course of business consistent with past practice and which, individually or in the aggregate, have not materially and adversely affected the condition (financial or otherwise), assets, operations, properties or business of either of the Acquiror or Matthews.

## ARTICLE VI REMEDIES FOR BREACHES OF THIS AGREEMENT

6.01. Survival of Representations, Warranties and Covenants. All of the representations, warranties and covenants of the parties contained in this Agreement shall survive the Closing (even if the damaged party knew or had

reason to know of any misrepresentation or breach of warranty at the time of the Closing) and shall continue in full force and effect thereafter for eighteen (18) months after the Closing Date; provided, however, that the representations and warranties in Section 5.01(m) hereof shall continue in full force and effect after the Closing Date until any claim with respect thereto is barred by the applicable statute of limitations.

6.02. Indemnification Provisions for Benefit of the Acquiror.

(a) If there is any material breach or inaccuracy of any of the representations, warranties or covenants of FDC, IEEC or the Shareholders contained herein, and provided that the Acquiror delivers to the Shareholders, as the case may be, pursuant to Section 6.04 hereof a claim for indemnification with respect to such alleged breach or inaccuracy, then the Shareholders, jointly and severally, shall indemnify the Acquiror from and against all Adverse Consequences that the Acquiror has suffered or may suffer caused by, resulting from, arising out of or relating to such breach or inaccuracy through and after the date of such claim.

(b) It is understood that the Shareholders are to be responsible for all Liabilities and obligations of FDC and IEEC, except those (i) disclosed on the Most Recent Balance Sheet (ii) incurred thereafter in the Ordinary Course of Business, (iii) incurred by FDC, IEEC or the Shareholders in connection herewith and expressly assumed in writing by the Acquiror, or (iv) set forth in the Disclosure Schedules. If the Acquiror is required to pay or otherwise discharge any Liability or obligation incurred by FDC and IEEC other than such a Liability, whether or not known to the Shareholders, FDC, IEEC or the Acquiror, and whether or not the existence of such Liability or obligation causes a breach or inaccuracy of any representation or warranty made in this Agreement, and provided that the Acquiror delivers to the Shareholders pursuant to Section 6.04 hereof a claim for indemnification with respect to such alleged Liability or obligation, then the Shareholders, jointly and severally, shall indemnify the Acquiror from and against all Adverse Consequences that the Acquiror has suffered or may suffer caused by, resulting from, arising out of or relating to such Liability or obligation.

(c) The Shareholders shall be obligated to indemnify as and to the extent set forth in this Agreement only if the aggregate of all claims by Acquiror and Matthews under such indemnity obligations exceeds \$50,000, it being understood that the \$50,000 amount is to serve as a "deductible" for the Shareholders' indemnification (for example, if the indemnity claims for which the Shareholders would, but for the

provisions of this Paragraph (c), be liable aggregate \$51,000, the Shareholders would then be liable for an aggregate of \$1,000, and not \$51,000).

6.03. Indemnification Provisions for Benefit of the Shareholders.

(a) If there is any material breach or material inaccuracy of any of the representations, warranties or covenants of the Acquiror or Matthews contained herein, and provided that the Shareholders deliver to the Acquiror pursuant to Section 6.04 hereof a claim for indemnification with respect to such alleged breach or inaccuracy, then the Acquiror and Matthews, severally, shall indemnify the Shareholders from and against all Adverse Consequences that the Shareholders have suffered or may suffer caused by, resulting from, arising out of or relating to such breach or inaccuracy through and after the date of such claim.

(b) The Acquiror and Matthews shall be obligated to indemnify as and to the extent set forth in this Agreement only if the aggregate of all claims by Shareholders under such indemnity obligations exceeds \$50,000, it being understood that the \$50,000 amount is to serve as a "deductible" for the Acquiror's and Matthews' indemnification (for example, if the indemnity claims for which the Acquiror and Matthews would, but for the provisions of this Paragraph (b), be liable aggregate \$51,000, and not \$51,000).

6.04. Notice of Claim for Indemnification. No claim for indemnification hereunder shall be valid unless notice of such claim is delivered to the Acquiror (in the case of a claim by the Shareholders) or to the Shareholders (in the case of a claim by the Acquiror or Matthews) within the applicable limitation period under Section 6.01. Any such notice shall set forth in

reasonable detail, to the extent known by the person giving such notice, the facts on which such claim is based and the estimated amount of Adverse Consequences resulting therefrom.

## 6.05. Matters Involving Third Parties.

(a) If Matthews, the Acquiror or the Shareholders receives notice or acquires knowledge of any matter which may give rise to a claim by another person and which may then result in a claim for indemnification under this Article VI, then (i) if such notice or knowledge is received or acquired by Matthews or the Acquiror, it shall promptly notify the Shareholders thereof, and (ii)if such notice or knowledge is received or acquired by the Shareholders, the Shareholders shall promptly notify the Acquiror thereof; provided, however, that no delay in giving such notice shall diminish any obligation under this Article VI to provide indemnification unless (and then solely to the extent that) the party from whom such indemnification is sought is prejudiced.

(b) Any party from whom such indemnification is sought (the "Indemnifying Party") shall have the right to defend the party seeking such indemnification (the "Indemnified Party") against such claim by another person (the "Third Party Claim") with counsel of the Indemnifying Party's choice reasonably satisfactory to the Indemnified Party so long as (I) within fifteen days after the Indemnified Party has given notice of the Third Party Claim to the Indemnifying Party, the Indemnifying Party notifies the Indemnified Party that the Indemnifying Party will indemnify the Indemnified Party from and against all Adverse Consequences the Indemnified Party may suffer caused by, resulting from, arising out of or relating to such Third Party Claim, (ii) the Indemnifying Party provides the Indemnified Party with evidence reasonably satisfactory to the Indemnified Party that the Indemnifying Party has the financial resources necessary to defend against the Third Party Claim and fulfill its indemnification obligations hereunder, (iii) the Third Party Claim seeks only money damages and not an injunction or other equitable relief, (iv) settlement of, or an, adverse judgment with respect to, the Third Party Claim is not, in the good faith judgment of the Indemnified Party, likely to establish a precedential custom or practice adverse to the continuing business interests of the Indemnified Party, and (v)the Indemnifying Party conducts the defense of the Third Party Claim actively and diligently.

(c) So long as the Indemnifying Party is conducting the defense of the Third Party Claim in accordance with Section 6.05(b) hereof, (i) the Indemnified Party may retain separate co-counsel at its sole cost and expense and participate in the defense of the Third Party Claim, (ii) the Indemnified Party shall not consent to the entry of any judgment or enter into any settlement with respect to the Third Party Claim without the prior consent of the Indemnifying Party, and (iii) the Indemnifying Party shall not consent to the entry of any judgment or enter into any settlement to the entry of any judgment or enter into any settlement with respect to the Third Party Claim without the prior consent to the entry of any judgment or enter into any settlement with respect to the Third Party Claim without the prior consent of the Indemnified Party.

(d) If any of the conditions specified in Section 6.05(b) hereof is or becomes unsatisfied, however, (i) the Indemnified Party, upon prior written notice to the Indemnifying Party, may defend against, and consent to the entry of any judgment or enter into any settlement with respect to, the Third Party Claim in any manner it may deem advisable (and the Indemnified Party need not consult with, or obtain any consent from, any Indemnifying Party in connection therewith), (ii) the Indemnifying Parties shall reimburse the Indemnified Party promptly and periodically for the costs of defending against the Third Party Claim (including reasonable attorneys' and accountants' fees and disbursements and amounts paid in settlement), and (iii) the Indemnifying Party shall remain responsible for any Adverse Consequences the Indemnified Party may suffer caused by, resulting from, arising out of or relating to such Third Party Claim to the fullest extent provided in this Article VI.

6.06. Treatment of Indemnification Payments. All indemnification payments made by the Shareholders, Matthews or the Acquiror under this Article VI shall be deemed adjustments to the Purchase Price.

6.07. Limitation on Amount of Indemnification. The maximum amount of liability for indemnification under this Article VI shall be (a) \$8,000,000 in the aggregate for the Shareholders and (b) \$8,000,000 in the aggregate for the Acquiror and Matthews.

6.08. Other Indemnification Provisions. Except as set forth in Section 7.09

hereof, the indemnification provisions in this Article VI are the sole and exclusive remedy of the parties hereto with respect to any breach of any representation, warranty or covenant set forth in this Agreement.

6.09. Set-off. Notwithstanding any other provision herein to the contrary, the Acquiror and Matthews shall be entitled, to the fullest extent permitted by Law, to set-off the liquidated amount of any indemnity obligations of the Shareholders under this Article VI, as well as the amount of any other Liabilities or obligations of the Shareholders to the Acquiror or Matthews, whether arising under this Agreement or otherwise, against any amounts payable by the Acquiror or Matthews to the Shareholders, whether arising under this Agreement or otherwise. In addition, if at any time that a payment is otherwise due from the Acquiror or Matthews to the Shareholders (whether arising under this Agreement or otherwise), a claim has been asserted or threatened by any Person which Acquiror believes in its reasonable judgment might give rise to an indemnity obligation on the part of the Shareholders hereunder, then the Acquiror or Matthews shall be entitled to withhold such payment (or such lesser portion as the Acquiror or Matthews deems appropriate) pending resolution of such claim. If an indemnity obligation actually arises as a result of such asserted or threatened claim, the Acquiror and Matthews shall be entitled to apply the amount of any withheld payment to the satisfaction of such indemnity obligation. The balance remaining, if any, shall be paid to the Shareholders.

## ARTICLE VII MISCELLANEOUS

7.01. Further Assurances. On and after the Closing Date, FDC, IEEC and the Shareholders, Matthews and the Acquiror shall take all appropriate action and execute all documents, instruments or conveyances of any kind which may be necessary or advisable to carry out any of the provisions hereof and to consummate the transactions contemplated hereby.

7.02. Expenses. The Acquiror or Matthews shall bear all expenses incurred by the Acquiror or Matthews in connection with the negotiation, preparation, execution and performance of this Agreement. The Shareholders shall bear all expenses incurred by FDC, IEEC and the Shareholders in connection with the negotiation, preparation, execution and performance of this Agreement.

7.03. Press Releases and Public Announcements. Upon execution hereof, the Acquiror shall cause to be made such press releases and public announcements as shall be required by law or deemed by the Acquiror to be necessary or desirable.

7.04. Notices. All notices, demands, claims, requests, undertakings, consents, opinions and other communications which may or are required to be given hereunder or with respect hereto shall be in writing, shall be given either by personal delivery or by mail, facsimile transmission (with confirmation of receipt), telegraph, telex or similar means of communication, and shall be deemed to have been given or made when delivered, if personally delivered, and otherwise when received, addressed to the respective parties as follows:

(a) If to the Acquiror or Matthews:

James L. Parker c/o Matthews International Corporation Two NorthShore Center Pittsburgh, PA 15212

With a copy to:

Reed Smith Shaw & McClay 435 Sixth Avenue P.O. Box 2009 Pittsburgh, PA 15230 Attention: Pasquale D. Gentile, Jr., Esq.

or to such other address as the Acquiror may from time to time designate by notice to FDC, IEEC and the Shareholders, made in accordance with the requirements of this Section 7.04 with respect to future notices, demands and other communications to the Acquiror;

(b) If to FDC, IEEC or the Shareholders:

Paul Rahill Industrial Equipment and Engineering Company, Inc. P.O. Box 547796 Orlando, FL 32854

and to:

Kenneth Robinson Industrial Equipment and Engineering Company, Inc. P.O. Box 547796 Orlando, FL 32854 Attention: Timothy J. Manor, Esquire

With a copy to:

Lowndes Drosdick Doster Kantor & Reed, P.A. 215 North Eola Drive Orlando, FL 32802 Attention: Timothy J. Manor, Esquire

or to such other address as FDC, IEEC or the Shareholders may from time to time designate by notice to the Acquiror made in accordance with the requirements of this Section 7.04 with respect to future notices, demands and other communications to FDC, IEEC or the Shareholders.

7.05. No Third-Party Beneficiaries. This Agreement shall not confer any rights or remedies upon any person other than Matthews, the Acquiror, FDC, IEEC, the Shareholders and their respective heirs, legal representatives, successors and permitted assigns.

7.06. Dispute Resolution.

(a) Specific Performance. Each of the parties to this Agreement acknowledges and agrees that other parties hereto would be damaged irreparably in the event any of the provisions of this Agreement are not performed in accordance with their specific terms or otherwise are breached. Accordingly, Matthews and the Acquiror agree that FDC, IEEC and the Shareholders, and FDC, IEEC and the Shareholders agree that the Acquiror and Matthews shall be entitled to an injunction or injunctions to prevent breaches of the provisions of this Agreement and to enforce specifically this Agreement and the terms and provisions hereof in any action instituted in any court of the United States or any state thereof having jurisdiction over the parties and the matter, in addition to any other remedy to which Matthews, the Acquiror, FDC, IEEC or the Shareholders may be entitled at law or in equity, but all subject to Section 7.06(c) hereof.

(b) Governing Law. This Agreement shall be governed by and construed in accordance with the Laws of Florida without giving effect to any choice or conflict of law provision or rule (whether of Florida or any other jurisdiction) that would cause the application of the Laws of any jurisdiction other than Florida.

7.07. Amendments and Waivers. No amendment of any provision of this Agreement, and no postponement or waiver of any such provision or of any default, misrepresentation or breach of warranty or covenant hereunder, whether intentional or not, shall be valid unless such amendment, postponement or waiver is in writing and signed by or on behalf of Matthews, the Acquiror, FDC, IEEC and the Shareholders. No such amendment, postponement or waiver shall be deemed to extend to any prior or subsequent matter, whether or not similar to the subject-matter of such amendment, postponement or waiver. No failure or delay on the part of Matthews, the Acquiror, FDC, IEEC or the Shareholders in exercising any right, power or privilege under this Agreement shall operate as a waiver thereof, nor shall any single or partial exercise of any right, power or privilege.

7.08. Succession and Assignment. This Agreement shall be binding upon and inure to the benefit of the parties named herein and their respective heirs,

legal representatives, successors and permitted assigns. No party may assign this Agreement or any of such party's rights, interests or obligations hereunder without the prior approval of the other parties hereto; provided, however, that Matthews and the Acquiror shall have the right to assign this Agreement or any of its rights, interests or obligations hereunder to any of its Affiliates without the prior approval of the other parties hereto.

### 7.09. Termination.

(a) Matthews, the Acquiror, FDC, IEEC and the Shareholders may terminate this Agreement as provided below in this Section 7.09(a):

(i) FDC, IEEC, the Shareholders, Matthews and the Acquiror may terminate this Agreement by mutual written consent at any time prior to the Closing;

(ii) The Acquiror or Matthews may terminate this Agreement by giving notice to FDC, IEEC and the Shareholders at any time prior to the Closing (A) if any representation, warranty or covenant of FDC, IEEC and the Shareholders contained in this Agreement is inaccurate or has been breached in any material respect, Matthews or the Acquiror has notified FDC, IEEC and the Shareholders of such inaccuracy or breach, and such inaccuracy or breach has continued without cure for a period of fifteen days after such notice, or (B)if the Closing shall not have occurred On or before April 30, 1996 because of the failure of any condition contained in Section 2.01 to be satisfied; or

(iii) Matthews or Acquiror may terminate this Agreement upon the receipt by IEEC of the notice referred to in Section 2.01(e) of this Agreement.

(iv) FDC, IEEC or the Shareholders may terminate this Agreement by giving notice to the Acquiror at any time prior to the Closing (A) if any representation, warranty or covenant of the Acquiror or Matthews contained in this Agreement is inaccurate or has been breached in any material respect, FDC, IEEC or the Shareholders have notified the Acquiror of such inaccuracy or breach, and such inaccuracy or breach has continued without cure for a period of fifteen days after such notice, or (B) if the Closing shall not have occurred on or before April 30, 1996 because of the failure of any condition contained in Section 2.02 to be satisfied.

(b) If this Agreement is terminated pursuant to Section 7.09(a) hereof, all rights and obligations of the parties hereunder shall terminate without any liability of any party to any other party provided, however, that (i) the provisions of Sections 3.04, 7.02 and this Section 7.09(b) shall remain in full force and effect, (ii) neither Matthews nor the Acquiror shall, for a period of two (2) years after the date of such termination, without the prior written approval of the Shareholders, FDC and IEEC, directly or indirectly solicit, encourage, entice or induce any senior employee of FDC or IEEC on the date hereof or at any time hereafter prior to such termination, to terminate his or her employment with FDC or IEEC, and (iii) termination pursuant to Section 7.02(a) shall not relieve any party of its liability (as specified in Section 7.09(c) hereof) for its willful breach of any material covenant or agreement of such party.

7.10. Matters of Construction, Interpretation and the Like.

(a) Construction. Matthews, the Acquiror, FDC, IEEC and the Shareholders have participated jointly in the negotiation and drafting of this Agreement. If an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by Matthews, the Acquiror, FDC, IEEC and the Shareholders and no presumption or burden of proof shall arise favoring or disfavoring either Matthews, the Acquiror or FDC, IEEC and the Shareholders because of the authorship of any of the provisions of this Agreement. Any reference to any federal, state, local or foreign statute or Law shall be deemed also to refer to all rules and regulations promulgated thereunder, unless the context requires otherwise. The word "including" shall mean including without limitation. Each representation, warranty and covenant contained herein shall have independent significance. If Matthews, the Acquiror, FDC, IEEC or the Shareholders breach in any respect any representation, warranty, covenant or other obligation contained herein or created hereby, the fact that there exists another representation, warranty, covenant or obligation relating to the same subject matter (regardless of the relative levels of specificity) which has not been breached shall not detract from or mitigate the consequences of such breach. The rights and remedies

expressly specified in this Agreement are cumulative and are not exclusive of any rights or remedies which any party would otherwise have except as expressly limited in this Agreement. The exhibits and schedules specified in this Agreement are incorporated herein by reference and made a part hereof. The article

and section headings hereof are for convenience only and shall not affect the meaning or interpretation of this Agreement.

(b) Severability. The invalidity or unenforceability of one or more of the provisions of this Agreement in any situation in any jurisdiction shall not affect the validity or enforceability of any other provision hereof or the validity or enforceability of the offending provision in any other situation or jurisdiction.

(c) Entire Agreement; Counterparts. This Agreement (including the documents referred to herein) constitutes the entire agreement among the parties and supersedes any prior understandings, agreements or representations by or among the parties, written or oral, to the extent they relate to the subject matter hereof. This Agreement may be executed in any number of counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument. It shall not be necessary in making proof of this Agreement to produce or account for more than one such counterpart.

### ARTICLE VIII DEFINITIONS

In addition to the terms defined elsewhere in this Agreement, the terms defined in this Article VIII shall, for all purposes of this Agreement, have the meanings specified or referred to in this Article VIII.

"Adverse Consequences" means all actions, suits, proceedings, hearings, investigations, charges, complaints, claims, demands, injunctions, judgments, orders, decrees, rulings, damages, dues, penalties, fines, costs, amounts paid in settlement, Liabilities, obligations, Taxes, Liens, losses, expenses and fees, including court costs and attorneys' and accountants' fees and disbursements.

"Affiliate" has the meaning set forth in Rule 12b-2 of the regulations promulgated under the Exchange Act.

"Affiliated Group" means any affiliated group within the meaning of Code Section 1504 or any similar group defined under a similar provision of state, local or foreign law.

"Agreement" or "this Agreement" means this instrument as originally executed and delivered, or, if amended or supplemented, as so amended or supplemented.

"Basis" means any past or present fact, situation, circumstance, status, condition, activity, practice, plan, occurrence, event, incident, action, failure to act or transaction that forms or could form the basis for any specified consequence.

"Closing" has the meaning specified in Section 1.1 0 hereof.

"Closing Date" means the date on which the Closing is completed.

"Code" means the Internal Revenue Code of 1986, as amended.

"Confidential Information" means any information not generally available to the public concerning the assets, liabilities, business, operations and/or prospects of FDC, IEEC, the Shareholders or the Acquiror.

"Disclosure Schedule" means the schedule, dated as of the date of this Agreement, setting forth various information relating to the representations and warranties of FDC, IEEC and the Shareholders in Article V hereof.

"Effective Time" has the meaning set forth in Section 1.04 hereof.

"Employee Benefit Plan" means any (a) nonqualified deferred compensation or retirement plan or arrangement which is an Employee Pension Benefit Plan, (b)

qualified defined contribution retirement plan or arrangement which is an Employee Pension Benefit Plan, (c) qualified defined benefit retirement plan or arrangement which is an Employee Pension Benefit Plan (including any Multi employer Plan), (d) Employee Welfare Benefit Plan or (e) other plan or practice of FDC or IEEC, whether formal or informal, written or oral, and whether or not legally enforceable, providing or which may provide benefits to the employees of FDC or IEEC in connection with their employment or the termination of their employment by retirement or otherwise.

"Employee Pension Benefit Plan" has the meaning set forth in ERISA Section 3(2).

"Employee Welfare Benefit Plan" has the meaning set forth in ERISA Section 3(1).

"Employment Agreements" means the Employment and Non-Competition Agreements to be executed and delivered by the Acquiror and the Shareholders in the form attached hereto as Exhibit A.

"Environmental, Health and Safety Laws" means the Comprehensive Environmental Response, Compensation and Liability Act of 1980, the Resource Conservation and Recovery Act of 1976, and the Occupational Safety and Health Act of 1970', each as amended, together with all other Laws of federal, state, local, and foreign governments (and all agencies thereof) concerning pollution or protection of the environment, public health and safety or employee health and safety, including Laws relating to emissions, discharges, releases or threatened releases of pollutants, contaminants or chemical, medical, industrial, hazardous or toxic materials or wastes

into ambient air, surface water, ground water or lands or otherwise relating to the manufacture, processing, distribution, use, treatment, storage, disposal, transport or handling of pollutants, contaminants or chemical, medical. industrial, hazardous or toxic materials or wastes.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended.

"Exchange Act" has the meaning set forth in Section 5.02(e) hereof.

"Extremely Hazardous Substance" has the meaning set forth in Section 302 of the Emergency Planning and Community Right-to-Know Act of 1986, as amended.

"Fiduciary" has the meaning set forth in ERISA Section 3(21).

"Financial Statements" means the various financial statements, including the reports of independent accountants, referred to in clause (I) of Section 5.01(f) hereof.

"Governmental Body" means any United States, state, county, city, municipal, regional, local or foreign organ of government, including all courts, boards and agencies of any thereof, and also includes all professional licensing or accrediting bodies with licensing, disciplinary or other powers relating to the practice of medicine.

"Indemnified Party" has the meaning set forth in Section 6.05(b) hereof.

"Indemnifying Party" has the meaning set forth in Section 6.05(b) hereof.

"Intellectual Property" means (a) all inventions (whether patentable or unpatentable and whether or not reduced to practice), all improvements thereto, and all patents, patent applications and patent disclosures, together with all reissuances, continuations, continuations-in-part, revisions, extensions and reexaminations thereof, (b) all trademarks, service marks, trade dress, logos, trade names and corporate names, together with all translations, adaptations, derivations and combinations thereof and including all goodwill associated therewith, and all applications, registrations and renewals in connection therewith, (c) all copyrightable works, all copyrights and all applications, registrations and renewals in connection therewith, (d) all mask works and all applications, registrations and renewals in connection therewith, (e) all trade secrets and confidential business information (including ideas, research and development, know-how, formulas, compositions, manufacturing and production processes and techniques, technical data, designs, drawings, specifications, customer and supplier lists, pricing and cost information and business and marketing plans and proposals), (f) all computer software (including data and related documentation), (g) all other

proprietary rights and (h) all copies and tangible embodiments thereof (in whatever form or medium).

"Internal Revenue Service" means the United States Internal Revenue Service.

"Knowledge", whether or not capitalized, means actual knowledge after reasonable investigation.

"Law" means statutes, rules, regulations, codes, plans, injunctions, judgments, orders, decrees and rulings of any Governmental Body.

"Lien" means any mortgage, pledge, encumbrance, charge or other security interest other than Permitted Liens.

"Liability," whether or not capitalized, means any liability (whether known or unknown, whether asserted or unasserted, whether absolute or contingent, whether accrued or unaccrued, whether liquidated or unliquidated, and whether due or to become due), including any liability for Taxes.

"Matthews' Exchange Act Filings" has the meaning set forth in Section 5.01 (aa) hereof.

"Matthews" has the meaning set forth in Section 1.03 hereof.

"Matthews Stock" has the meaning set forth in Section 1.03 hereof.

"Merger" has the meaning set forth in the Section 1.01 hereof.

"Most Recent Balance Sheet" means the balance sheets of FDC and IEEC dated as of December 31, 1995 and included in the Financial Statements.

"Multi employer Plan" means any employee benefit plan which is a "Multi employer plan" within the meaning of Section 4001 (a)(3) of ERISA and to which FDC or IEEC has an obligation to contribute.

"Ordinary Course of Business," whether or not capitalized, means the ordinary course of business of FDC, IEEC or the Acquiror, as the case may be, consistent with its past custom and practice.

"PBGC" means the Pension Benefit Guaranty Corporation.

"Permitted Liens" means as of any given time (a) easements, rights of way, title exceptions and reservations, restrictions, zoning ordinances and other encumbrances which do not adversely affect the use of the properties subject thereto by FDC or IEEC

in the Ordinary course of Business; and (b) obligations and duties of FDC or IEEC, not interfering with the Ordinary Course of Business, as tenant or subtenant under any leases of real or personal property wherever situated which are described in Section 5.01(g)(ii) of the Disclosure Schedule.

"Person," whether or not capitalized, means an individual, corporation, partnership, unincorporated organization, voluntary association, joint stock company, trust, joint venture or Governmental Body.

"Plan of Merger" has the meaning set forth in Section 1.09 hereof.

"Prohibited Transaction" has the meaning set forth in ERISA Section 406 and Code Section 4975.

"Purchase Price" means the net amount of cash and stock delivered to the Shareholders by the Acquiror as consideration for the Merger pursuant to Section 1.03 hereof, as such amount may be adjusted pursuant to Section 6.06 hereof.

"Real Property" has the meaning set forth in Section 5.01 (g)(i) hereof.

"Reportable Event" has the meaning set forth in ERISA Section 4043.

"Securities Act" shall mean the Securities Act of 1933, as amended, or any similar federal statute then in effect, and a reference to a particular section thereof shall be deemed to include a reference to the comparable section, if any, of any such similar federal statute.

"Shareholders" has the meaning set forth in the introduction hereof.

"Shares" means the shares of Common Stock, par value \$ per share, of FDC and IEEC owned by the shareholders.

"Subsidiary" means a corporation controlling, controlled by or under common control with FDC or IEEC as the case may be.

"Surviving Corporation" has the meaning set forth in Section 1.02 hereof.

"Tax" or "Taxes" means any federal, state, local, or foreign income, gross receipts, license, payroll, employment, excise, severance, stamp, occupation, premium, environmental (including taxes under Code Section 59A), customs duties, capital stock, franchise, profits, withholding, social security (or similar), unemployment, disability, real property, personal property, sales, use, transfer, registration, value added, alternative minimum, estimated or other tax of any kind whatsoever, including any interest, penalty or addition thereto, whether disputed or not.

"Tax Return" means any return, declaration, report, claim for refund or information return or statement relating to Taxes, including any schedule or attachment thereto, and including any amendment thereof.

"Third-Party Claim" has the meaning set forth in Section 6.05(b) hereof.

"Warn Act" means the Worker Adjustment and Retraining Notification Act of 1988, and the rules and regulations promulgated thereunder.

WITNESS the due execution of this Agreement and Plan of Merger as of the day and year first above written.

FLORIDA DEVELOPMENT	INDUSTRIAL EQUIPMENT AND	
CORPORATION	ENGINEERING COMPANY, INC.,	
a Delaware corporation		

P. Rahill By:

By: D.J. DeCarlo -----President Vice President

INDUSTRIAL EQUIPMENT AND ENGINEERING COMPANY, INC.,

a Florida corporation P. Rahill

MATTHEWS INTERNATIONAL CORPORATION, a Pennsylvania corporation

President

By:

By: D.J. DeCarlo

President, Bronze Division

P. Rahill

PAUL RAHILL, as a Shareholder

K.Robinson

KENNETH ROBINSON. as a Shareholder

# EXHIBIT A

### EMPLOYMENT AGREEMENT

THIS AGREEMENT, made as of the \_\_\_\_\_ day of March 1996, by and among INDUSTRIAL EQUIPMENT AND ENGINEERING COMPANY, INC., a Delaware corporation (the "Company"), MATTHEWS INTERNATIONAL CORPORATION, a Pennsylvania Corporation ("Matthews"), and Paul Rahill, of Apopka, Florida ("Employee").

#### WITNESSETH:

WHEREAS, the Company is engaged in the business of manufacturing and sale of equipment used for cremation and is a wholly-owned subsidiary of Matthews;

WHEREAS, the Company purchased the business from Employee pursuant to an Agreement and Plan of Merger dated March 5, 1996 (the "Merger Agreement") among Florida Development Corporation, a Florida corporation, Industrial Equipment and Engineering Company, Inc., a Florida corporation ("IEEC"), the Company, the Employee and Kenneth Robinson, and this Employment Agreement is being entered into in connection therewith;

WHEREAS, Employee possesses valuable knowledge and skills that will contribute to the successful operation of the Company's business; and

WHEREAS, the Company and Employee have agreed to execute and deliver this Agreement in consideration, among other things, of (i) the access Employee has had and will continue to have to confidential or proprietary information of the Company, (ii) the access Employee will have to confidential or proprietary information to be acquired hereafter by the Company, (iii) the willingness of the Company to make valuable benefits available hereafter to Employee, and (iv) Employee's receipt of compensation from time to time by the Company; and

WHEREAS, the Company desires to procure the services of Employee, and Employee is willing to enter the employment of the Company, upon the terms and subject to the conditions hereinafter set forth;

WHEREAS, Matthews desires to enter into this Agreement as it believes it will benefit from Employee's services to the Company and by the Company's contributions to Matthews operations;

NOW, THEREFORE, intending to be legally bound, the Company agrees to employ Employee, and Employee hereby agrees to be employed by the Company, upon the following terms and conditions:

### ARTICLE I EMPLOYMENT

1.01. Office. Employee is hereby employed as President of the Company and in such capacity shall use his best energies and abilities in the performance of his duties as the chief executive of the Company hereunder and in the performance of such other chief executive duties as may be assigned to him from time to time by the Board of Directors of the Company. Except for reasonable travel in the ordinary course of business, Employee shall not be required to relocate from the Orlando, Florida area to perform his duties hereunder. The Company shall furnish the Employee with an office, a secretary, and such other facilities and services suitable to his position and adequate for the performance of his duties hereunder.

1.02. Term. Subject to the terms and provisions of Article II hereof, Employee's employment hereunder shall commence as of March \_\_, 1996 and shall continue until March \_\_, 1999, unless extended by subsequent agreement of the parties hereto; provided, however, that the provisions of Section 1.04(b) hereof shall continue in full force and effect until all amounts payable thereunder are received by Employee.

1.03. Base Salary. During the term of Employee's employment hereunder, compensation shall be paid to Employee by the Company at the rate of \$110,000

per annum (the "Base Salary "), payable no less frequently than monthly. The rate of compensation to be paid to Employee may be adjusted upward or downward (but not below the Base Salary) by the Board of Directors of the Company at any time based upon Employee's contribution to the success of the Company as measured by the Hay system and overall salary administration guidelines used by Matthews, and on such other factors as the Board of Directors of the Company shall deem appropriate. A brief description of the Hay system is attached as Exhibit A.

1.04. Additional Incentive Compensation.

(a) Incentive Compensation Plan. In lieu of participation in any incentive plan offered by the Company or Matthews (except as set forth in Section 1.04(b) below), Employee shall be entitled to participate in the Incentive Compensation Plan attached hereto as Exhibit B, with a maximum payout target of 30 % of Base Salary.

(b) Growth Participation Plan. In addition to the incentive compensation provided in Section 1.04(a), the Employee shall be entitled to the amounts set forth in this Section 1.04(b).

(i) The Company shall pay to Employee additional compensation under its "Growth Participation Plan" created hereby, at the conclusion of the five-year period (the "Five-Year Measuring Period") that commences on April 1, 1996 (the "Commencement Date"). Subject to interim payments described in subsection (ii) below, any such additional compensation shall be paid to the Employee on or before June 1, 2001. The total amount of additional compensation payable to Employee under the Growth Participation Plan shall be \$100,000 when EBIT (as defined in subsection [iii] below) exceeds \$8,000,00 and 50% of the amount by which EBIT exceeds \$8,000,000, but does not exceed \$10,600,000. For example, EBIT of \$8,700,000 will entitle employee to \$450,000 under such Growth Participation Plan. To the extent EBIT exceeds \$10,600,000, but does not exceed \$16,000,000, Employee shall receive in addition to the amounts aforesaid 45% of the EBIT in excess of \$10,600,000 net of tax calculated using the Company's effective tax rate. No compensation shall be payable under the Growth Participation Plan with respect to that portion of EBIT which is less than \$8,000,000 or more than \$16,000,000.

(ii) Interim payments against the ultimate Growth Participation Plan compensation to be paid at the end of the Five-Year Measuring Period shall be made by the Company as follows: Within sixty days of each anniversary of the Commencement Date, the Company shall determine the extent (if any) to which the projected cumulative earnings of the Company before tax since the Commencement Date, calculated in the same manner as EBIT in Section 1.04(b)(iii) would qualify for additional incentive compensation under the provisions of subsection 1.04(b)(i) above, if continued at the same rate until the conclusion of the Five Year Measuring Period. Such projection shall be calculated on the basis of the historical growth rate experienced from the Commencement Date through the end of the year for which the calculation is being made. The Employee shall be entitled to receive at each such annual determination, as an advance, ten percent of the amount which is thereby projected to be payable as of the conclusion of the Five Year Measuring Period. Any such payments shall be credited against the aggregate ultimate Growth Participation Plan incentive compensation due to Employee at the end of the Five-Year Measuring Period as set forth in subsection (i) above. Any such payment shall be made within sixty days after each anniversary of the Commencement Date.

(iii) For purposes of this Section 1.04(b), the "EBIT" of the Company shall mean the total consolidated gross revenues of the Company for the Five Year Measuring Period, less all operating expenses of the Company, but (1) not including any gains resulting from the sale, conversion or other disposition of capital assets (i.e., property, plant and equipment) or any gains resulting from the write-up of assets, (2) not including in such expenses and charges any losses resulting from the sale, conversion or other disposition of capital assets, any amortization of the Company's goodwill, any interest paid on long-term debt of the Company, any allocations for overhead costs attributable to Matthews (as opposed to the Company), any depreciation or amortization resulting from the write-up of assets, or any payments

made under the Growth Participation Plan described in this Section 1.04(b), (3) not including the effect of any current or deferred federal or state income

taxes, and (4) including depreciation (other than depreciation on the write-up of assets) and an imputed expense of \$64,000 per year which represents the imputed fair market value of the rental of the real estate upon which the Company operates; all determined in accordance with generally accepted accounting principles applied consistently with those applied by Matthews.

(c) Covenant to Continue Business. Each of the Company and Matthews agrees that, at least through June 1, 2001, the Company's operations and business shall be continued in substantially the same manner as conducted by IEEC prior to its purchase by the Company pursuant to the Merger Agreement. In furtherance of such business and operations, each of the Company and Matthews agrees not to sell, convert or otherwise dispose of material capital assets (i.e., property, plant and equipment) of the Company without the prior written consent of Employee, and each of the Company and Matthews agrees that the Company shall not incur material obligations, including, without limitation, long-term debt, without the prior written consent of Employee.

(d) In the event that Matthews makes any acquisition of, or merges directly or indirectly with, any designer, manufacturer or seller of cremation equipment or related crematory accessories, parts or supplies subsequent to the date hereof, Matthews agrees that such business will be owned by the Company and become part of the business previously operated by Industrial Equipment and Engineering Co. (a Florida corporation), unless Employee objects in writing to such treatment within five (5) business days of receiving notice of the terms and conditions of such acquisition or merger in substantially final form. For purposes of Section 1.04(b), the purchase price of such acquisition or merger not objected to by Employee will be treated as an operating expense of the Company charged at a cost of capital of 13% or such other percentage as may be agreed by the parties.

1.05. Employee Benefits. At all times during the term of Employee's employment hereunder, Employee shall (a) be entitled to an automobile allowance of up to \$600 per month consistent with the past practice of Industrial Equipment & Engineering Co. (Florida), or, at the Employee's option, be entitled to the automobile benefits otherwise provided to those executives of Matthews who are at a comparable grade level and (b) be covered by such major medical or health benefit plans and pensions as are available generally to employees of the Company.

1.06. Expense Reimbursements. Upon the submission by Employee of supporting documentation, the Company shall reimburse the Employee for all reasonable expenses actually paid or incurred by Employee in the course and pursuant to the business of the Company including expenses for travel and entertainment.

### ARTICLE II TERMINATION

2.01. Illness, Incapacity. If during the term of Employee's employment hereunder Employee shall be prevented, in the Company's judgment, from effectively performing all his duties hereunder by reason of illness or disability for a period of one hundred and twenty (120) days during any twelve month period, then the Company may, by written notice to Employee, terminate Employee's employment hereunder. Upon delivery to Employee of such notice, together with payment of any salary accrued under Section 1.03 hereof and any disability benefits payable to Employee pursuant to disability plans generally available to employees of the Company, Employee's employment and all obligations of the Company under Article 1 hereof shall forthwith terminate, other than the obligations under Section 1.04(b), which shall remain in effect. The obligations of Employee under Article IV hereof shall continue notwithstanding termination of Employee's employment pursuant to this Section 2.01.

2.02. Death. If Employee dies during the term of his employment hereunder, Employee's employment hereunder shall terminate and all obligations of the Company hereunder shall terminate, other than any obligations with respect to the payment of accrued and unpaid salary, any death benefits payable to Employee pursuant to death benefit plans generally available to employees of the Company and the obligations under Section 1.04(b).

2.03. Company Termination. If Employee (a) willfully and materially breaches the terms of the Agreement and does not cure such breach within thirty (30) days after receipt by Employee of Notice of same, (b) commits fraud or

embezzlement against the Company, or (c) is convicted of committing a crime which is a felony, then, and in any such event ("Cause"), the Company may, by written notice to Employee, terminate Employee's employment hereunder. Upon delivery to Employee of such notice, together with payment of any salary accrued under Section 1.03 hereof, Employee's employment and all obligations of the Company under Article I hereof shall forthwith terminate. The obligations of Employee's employment pursuant to this Section 2.03. If the Company terminates the Employee for any reason other than Cause, the Employee shall be entitled to receive the Base Salary through March 31, 1999, and shall be entitled to receive all compensation payable pursuant to Sections 1.04 and 1.05 hereof through March 31, 1999 (and, in the case of the Growth Participation Plan, through April 1, 2001).

2.04. Employee Termination. Employee agrees to give the Company sixty (60) days prior written notice of the termination of his employment with the Company. Simultaneously with such notice, Employee shall inform the Company in writing as to his employment plans following the termination of his employment with the Company. Upon the termination of his employment all obligations of the Company under Article

I hereof shall terminate. The obligations of Employee under Article IV hereof shall continue notwithstanding termination of Employee's employment pursuant to this Section 2.04.

2.05. Termination Upon Change in Business. If (a) this Agreement is terminated upon the insolvency or bankruptcy of the Company, or upon a discontinuance of operations of the Company by Matthews, (b) this Agreement is assigned in connection with the sale of all or substantially all of the assets of the Company, or Matthews sells all of the capital stock of the Company, then, in any such case, and notwithstanding anything to the contrary in this Agreement, (I) Employee may terminate his employment hereunder under Section 2.04 and the provisions of Section 4.05 hereof shall not apply and (ii) Employee shall not forfeit his right to payments under Section 1.04(b) hereof.

# ARTICLE III EMPLOYEE'S ACKNOWLEDGMENTS

Employee recognizes and acknowledges that: (a) in the course of Employee's employment by the Company it will be necessary for Employee to acquire information which could include, in whole or in part, information concerning the Company's sales, sales volume, sales methods, sales proposals, customers and prospective customers, identity of customers and prospective customers, identity of key purchasing personnel in the employ of customers and prospective customers, amount or kind of customer's purchases from the Company, the Company's sources of supply, the Company's computer programs, system documentation, special hardware, product hardware, related software development, the Company's manuals, formulae, processes, methods, machines, compositions, ideas, improvements, inventions or other confidential or proprietary information belonging to the Company or relating to the Company's affairs (collectively referred to herein as the "Confidential Information"); (b) the Confidential Information is the property of the Company; (c) the use, misappropriation or disclosure of the Confidential Information would constitute a breach of trust and could cause irreparable injury to the Company; and (d) it is essential to the protection of the Company's good will and to the maintenance of the Company's competitive position that the Confidential Information be kept secret and that Employee not disclose the Confidential Information to others or use the Confidential Information to Employee's own advantage or the advantage of others.

Employee further recognizes and acknowledges that it is essential for the proper protection of the business of the Company that Employee be restrained (a) from soliciting or inducing any employee of the Company to leave the employ of the Company, (b) from hiring or attempting to hire any employee of the Company, (c) from soliciting the trade of or trading with the customers and suppliers of the Company for any business purpose, and (d) from competing against the Company for a reasonable

period (two [2] years) following the termination of Employee's employment with the Company.

#### ARTICLE IV EMPLOYEE'S COVENANTS AND AGREEMENTS

4.01. Non-Disclosure of Confidential Information. Employee agrees to hold and safeguard the Confidential Information in trust for the Company, its successors and assigns and agrees that he shall not, without the prior written consent of the Company, misappropriate or disclose or make available to anyone for use outside the Company's organization at any time, either during his employment with the Company or subsequent to the termination of his employment with the Company for any reason, including without limitation termination by the Company for cause, any of the Confidential Information, whether or not developed by Employee, except as required in the performance of Employee's duties to the Company.

4.02. Disclosure of Works and Inventions/Assignment of Patents. Employee shall disclose promptly to the Company or its nominee any and all works, inventions, discoveries and improvements authored, conceived or made by Employee during the period of employment and related to the business or activities of the Company, and hereby assigns and agrees to assign all his interest therein to the Company or its nominee. Whenever requested to do so by the Company, Employee shall execute any and all applications, assignments or other instruments which the Company shall deem necessary to apply for and obtain Letters Patent or Copyrights of the United States or any foreign country or to otherwise protect the Company's interest therein. Such obligations shall continue beyond the termination of employment with respect to works, inventions, discoveries and improvements authored, conceived or made by Employee during the period of employment, and shall be binding upon Employee's assigns, executors, administrators and other legal representatives.

4.03. Duties. Employee agrees to be a loyal employee of the Company. Employee agrees to devote his best efforts full time consistent with past practice to the performance of his duties for the Company, to give proper time and attention to furthering the Company's business, and to comply with all rules, regulations and instruments established or issued by the Company and consistent with past practice. Employee further agrees that during the term of this Agreement, Employee shall not, directly or indirectly, engage in any business which would detract from Employee's ability to apply his best efforts to the performance of his duties hereunder. Employee also agrees that he shall not usurp any corporate opportunities of the Company.

4.04. Return of Materials. Upon the termination of Employee's employment with the Company for any reason, including without limitation termination by the Company for cause, Employee shall promptly deliver to the Company all correspondence, drawings, blueprints, manuals, letters, notes, notebooks, reports,

flowcharts, programs, proposals and any documents concerning the Company's customers or concerning products or processes used by the Company and, without limiting the foregoing, will promptly deliver to the Company any and all other documents or materials containing or constituting Confidential Information.

4.05. Restrictions on Competition. Unless either the Company or Matthews materially breaches this Agreement, Employee covenants and agrees that during the period of Employee's employment hereunder and for a period of two (2) years following the termination of Employee's employment, including without limitation termination by the Company for cause, but not including termination under Section 2.05 hereof, Employee shall not, in the United States of America, or in any other country of the world in which the Company has done business at any time during the last two (2) years prior to termination of Employee's employment with the Company, engage, directly or indirectly, whether as principal or as agent, officer, director, employee, consultant, shareholder, or otherwise, alone or in association with any other person, corporation or other entity, in any Competing Business. For purposes of this Agreement, the term "Competing Business" shall mean any person, corporation or other entity which sells or attempts to sell any products or services which are the same as or similar to the products and services sold by the Company at any time and from time to time during the last two (2) years prior to the termination of Employee's employment hereunder.

4.06. Non-Solicitation of Customers and Suppliers. Unless either the Company or Matthews materially breaches this Agreement, Employee agrees that during his employment with the Company he shall not, directly or indirectly, solicit the

trade of, or trade with, any customer, prospective customer, supplier, or prospective supplier of the Company for any business purpose other than for the benefit of the Company. Employee further agrees that for two (2) years following termination of his employment with the Company, including without limitation termination by the Company for cause, Employee shall not, directly or indirectly, solicit the trade of, or trade with, any customers or suppliers, or prospective customers or suppliers, of the Company in any Competing Business.

4.07. Non-Solicitation of Employees. Unless either the Company or Matthews materially breaches this Agreement, Employee agrees that, during his employment with the Company and for two (2) years following termination of Employee's employment with the Company, including without limitation termination by the Company for cause, Employee shall not, directly or indirectly, solicit or induce, or attempt to solicit or induce, any employee of the Company to leave the Company for any reason whatsoever, or hire any employee of the Company for any Competing Business.

### ARTICLE V EMPLOYEE'S REPRESENTATIONS AND WARRANTIES

5.01. No Prior Agreements. Employee represents and warrants that he is not a party to or otherwise subject to or bound by the terms of any contract, agreement or understanding which in any manner would limit or otherwise affect his ability to perform his obligations hereunder, including without limitation any contract, agreement or understanding containing terms and provisions similar in any manner to those contained in Article IV hereof. Employee further represents and warrants that his employment with the Company will not require him to disclose or use any confidential information belonging to prior employers or other persons or entities.

5.02. Employee's Abilities. Employee represents that his experience and capabilities are such that the provisions of Article IV will not prevent him from earning his livelihood, and acknowledges that it would cause the Company serious and irreparable injury and cost if Employee were to use his ability and knowledge in competition with the Company or to otherwise breach the obligations contained in Article IV.

5.03. Remedies. In the event of a breach by Employee of the terms of this Agreement, the Company shall be entitled, if it shall so elect, to institute legal proceedings to obtain damages for any such breach, or to enforce the specific performance of this Agreement by Employee and to enjoin Employee from any further violation of this Agreement and to exercise such remedies cumulatively or in conjunction with all other rights and remedies provided by law. Employee acknowledges, however, that the remedies at law for any breach by him of the provisions of this Agreement may be inadequate and that the Company shall be entitled to injunctive relief against him in the event of any breach.

### ARTICLE VI MISCELLANEOUS

6.01. Authorization to Modify Restrictions. It is the intention of the parties that the provisions of Article IV hereof shall be enforceable to the fullest extent permissible under applicable law, but that the unenforceability (or modification to conform to such law) of any provision or provisions hereof shall not render unenforceable, or impair, the remainder thereof. If any provision or provisions hereof shall be deemed invalid or unenforceable, either in whole or in part, this Agreement shall be deemed amended to delete or modify, as necessary, the offending provision or provisions and to alter the bounds thereof in order to render it valid and enforceable.

6.02. Tolling Period. The noncompetition, nondisclosure and non-solicitation obligations contained in Article IV hereof shall be extended by the length of time during which Employee shall have been in breach of any of the provisions of such Article IV.

6.03. Entire Agreement. This Agreement represents the entire agreement of the parties and may be amended only by a writing signed by each of them.

6.04. Governing Law. This Agreement shall be governed by and construed in accordance with the laws of the State of Florida.

6.05. Remedies. If the Employer prevails in a proceeding for damages or injunctive relief, Employee agrees that Employer, in addition to other relief, shall be entitled to reasonable attorneys' fees, costs and the expenses of litigation incurred by Employer in securing the relief granted by the Court.

6.06. Agreement Binding. The obligations of Employee under this Agreement shall continue after the termination of his employment with the Company for any reason, with or without cause, and shall be binding on his heirs, executors, legal representatives and assigns and shall be binding upon and inure to the benefit of any successors and assigns of the Company.

6.07. Counterparts, Section Headings. This Agreement may be executed in any number of counterparts, each of which shall be deemed to be an original, but all of which together shall constitute one and the same instrument. The section headings of this Agreement are for convenience of reference only and shall not affect the construction or interpretation of any of the provisions hereof.

IN WITNESS WHEREOF, the parties hereto have executed this Agreement or caused this Agreement to be executed the day and year first above written.

Witness:

- -----Name: \_\_\_\_\_

------PAUL RAHILL

Address:

- -----Name:

-----

- ------

INDUSTRIAL ENGINEERING AND EQUIPMENT COMPANY, INC., a Delaware Corporation

By:

Name:

\_\_\_\_\_ Name: ------Title:

Name:

------

MATTHEWS INTERNATIONAL CORPORATION, a Pennsylvania corporation

By:

Name:

Name:

------

Title:

-		
N	lame:	

# MATTHEWS INTERNATIONAL CORPORATION AND SUBSIDIARIES EXHIBIT 11 - COMPUTATION OF EARNINGS PER SHARE FOR THE SIX MONTHS ENDED MARCH 31, 1996 AND 1995

-----

	1996 1995
1. Net income	\$11,622,034 \$7,990,293
2. Weighted average number of common shares outstanding during the period	8,867,762 8,850,350
<ol> <li>Shares issuable upon exercise of dilutive stock options outstanding during period, based on higher of average or period-end values</li> </ol>	183,656 -
<ol> <li>Weighted average number of common shares outstanding during the period, assuming full dilution (2 + 3)</li> </ol>	9,051,418 8,850,350
5. Primary earnings per share (1 divided by 2)	\$ 1.31 \$ .90
<ol> <li>Fully diluted earnings per sha (1 divided by 4)</li> </ol>	re \$ 1.28 \$ .90

## <ARTICLE> 5 <LEGEND> THIS SCHEDULE CONTAINS SUMMARY FINANCIAL INFORMATION EXTRACTED FROM THE REGISTRANT'S QUARTERLY REPORT ON FORM 10-Q FOR THE SIX-MONTH PERIOD ENDED MARCH 31, 1996 AND IS QUALIFIED IN ITS ENTIRETY BY REFERENCE TO SUCH FINANCIAL STATEMENTS. </LEGEND>

<s> &lt;(</s>	1~
<period-type></period-type>	6-MOS
<fiscal-year-end></fiscal-year-end>	
<period-end></period-end>	MAR-31-1996
<cash></cash>	\$ 14,237,869
<securities></securities>	3,020,128
<receivables></receivables>	27,280,451
<allowances></allowances>	0
<inventory></inventory>	11,956,736
<current-assets></current-assets>	57,462,100
<pp&e></pp&e>	63,072,483
<depreciation></depreciation>	26,065,097
<total-assets></total-assets>	149,768,242
<current-liabilit< td=""><td>IES&gt; 22,980,100</td></current-liabilit<>	IES> 22,980,100
<bonds></bonds>	0
<preferred-manda< td=""><td>ATORY&gt; 0</td></preferred-manda<>	ATORY> 0
<preferred></preferred>	0
<common></common>	9,064,212
<other-se></other-se>	92,986,750
<total-liability-a< td=""><td>ND-EQUITY&gt; 149,768,242</td></total-liability-a<>	ND-EQUITY> 149,768,242
<sales></sales>	83,976,824
<total-revenues></total-revenues>	83,976,824
<cgs></cgs>	46,421,717
<total-costs></total-costs>	46,421,717
<other-expenses></other-expenses>	0
<loss-provision></loss-provision>	0
<interest-expense< td=""><td></td></interest-expense<>	
<income-pretax></income-pretax>	20,050,818
<income-tax></income-tax>	8,428,784
<income-continui< td=""><td>NG&gt; 11,622,034</td></income-continui<>	NG> 11,622,034
<discontinued></discontinued>	0
<extraordinary></extraordinary>	0
<changes></changes>	0
<net-income></net-income>	11,622,034
<eps-primary></eps-primary>	1.31
<eps-diluted></eps-diluted>	1.28

</TABLE>